Washington, DC 20375-5320



NRL/MR/6353--20-10,057

Metamaterials Using AdditiveManufacturing Technologies

ROBERT SAUNDERS

Multifunctional Materials Branch Materials Science and Technology Division

June 29, 2020

DISTRIBUTION STATEMENT A: Approved for public release; distribution is unlimited.

REPORT DOCUMENTATION PAGE

Form Approved OMB No. 0704-0188

Public reporting burden for this collection of information maintaining the data needed, and completing and review suggestions for reducing this burden to Department of D Suite 1204, Arlington, VA 22202-4302. Respondents sho information if it does not display a currently valid OMB cc	is estimated to average 1 ho ving this collection of informa efense, Washington Headqu ould be aware that notwithsta ontrol number. PLEASE DO	ur per response, including the ation. Send comments regard uarters Services, Directorate anding any other provision of NOT RETURN YOUR FORM	ne time for reviewing ins ding this burden estimate for Information Operatio f law, no person shall be M TO THE ABOVE ADD	tructions, searching existing data sources, gathering and e or any other aspect of this collection of information, including ons and Reports (0704-0188), 1215 Jefferson Davis Highway, subject to any penalty for failing to comply with a collection of DRESS.			
1. REPORT DATE (<i>DD-MM-YYYY</i>) 29-06-2020	2. REPORT TYPE NRL Memorand	dum Report	3	. DATES COVERED (From - To) Sept Oct. 2019			
4. TITLE AND SUBTITLE			5	a. CONTRACT NUMBER			
Metamaterials Using Additive Manufac	cturing Technologies		5	b. GRANT NUMBER			
				c. PROGRAM ELEMENT NUMBER			
6. AUTHOR(S)			5	d. PROJECT NUMBER			
Robert Saunders			5	e. TASK NUMBER			
			5	f. WORK UNIT NUMBER 1G76			
7. PERFORMING ORGANIZATION NAMI	E(S) AND ADDRESS	(ES)	8	. PERFORMING ORGANIZATION REPORT NUMBER			
Naval Research Laboratory 4555 Overlook Avenue, SW Washington, DC 20375-5320			NRL/MR/635320-10,057				
9. SPONSORING / MONITORING AGEN	CY NAME(S) AND AI	DDRESS(ES)	1	0. SPONSOR / MONITOR'S ACRONYM(S)			
Office of Naval Research				ONR 6.1			
875 N. Randolph Street, Suite 1425 Arlington, VA 22203-1995			1. SPONSOR / MONITOR'S REPORT NUMBER(S)				
12. DISTRIBUTION / AVAILABILITY STATEMENT DISTRIBUTION STATEMENT A: Approved for public release; distribution is unlimited.							
13. SUPPLEMENTARY NOTES							
14. ABSTRACT			1 . 11 . 1				
The purpose of this document is to provide an overview of current research being conducted in the field of additively manufactured metamaterials. This includes a historical overview of both metamaterials and additive manufacturing as well as suggestions for future research directions.							
15. SUBJECT TERMS							
Additive manufacturing Metamaterials 4D materials							
16. SECURITY CLASSIFICATION OF:	17. LIMITATION OF ABSTRACT	18. NUMBER OF PAGES	19a. NAME OF RESPONSIBLE PERSON Robert Saunders				
a. REPORTb. ABSTRACTUnclassifiedUnclassifiedUnlimitedUnlimited	c. THIS PAGE Unclassified Unlimited	Unclassified Unlimited	35	19b. TELEPHONE NUMBER (include area code) (202) 404-3489			

This page intentionally left blank.

CONTENTS

EX	ECUTIVE SUMMARY	E-1
1.	INTRODUCTION	1
2.	 ADDITIVE MANUFACTURING AND METAMATERIALS 2.1 Additive Manufacturing Technologies 2.2 Metamaterials 	1 2 3
3.	 FABRICATION AND ADDITIVE MANUFACTURING OF METAMATERIAL	5 5 7
4.	 ADDITIVELY MANUFACTURED METAMATERIALS	8 9 11
5.	SUMMARY AND FUTURE WORK	13
AC	CKNOWLEDGMENTS	15
RE	FERENCES	15

FIGURES

1	Publications per year	1
2	AM process classification	2
3	Electromagetic metamaterials	4
4	Acoustic metamaterials	5
5	Mechanical metamaterials	6

EXECUTIVE SUMMARY

The purpose of this report is to provide an overview and summary of existing literature on the use of additive manufacturing (AM) technologies to fabricate metamaterials. The areas of both AM and metamaterials have exploded in popularity in the last two decades with each having tens of thousands of publications in that time. AM, or 3D printing, is the process of joining materials to make parts from 3D model data, usually layer upon layer, as opposed to subtractive manufacturing and formative manufacturing methodologies. The process of additive manufacturing is very flexible in both part design and material selection. Metamaterials are fabricated structures and composite materials that either mimic known material responses or qualitatively have new, physically realizable response functions that do not occur or may not be readily available in nature. Metamaterials were originally theorized and developed for electromagnetic applications but have since been extended to optical, thermal, acoustic, and mechanical applications. Metamaterials are commonly said to be "artificial" materials since the exotic properties that c an be demonstrated d o not depend on chemical constituents, but rather internal structure of the metamaterial. As expected, this internal structure can be complex and difficult to fabricate. The freedom derived from AM is therefor a natural choice for fabricating metamaterials.

In order to achieve the previously unachievable behaviors seen in metamaterials, the internal structure of the metamaterial manipulates wave physics at the scale of the wave length. Therefor, depending on the application, the internal structure needs to be anywhere from millimeters to nanometers in scale. Early fabrication methods were based on lithography as it was able to achieve structure resolutions at the submicron scale. However, lithography based methods generally suffer from a number of limitations such as difficulty in large area 3D production, the use of "negatives", and low s peed. With advances in AM, it has became increasingly more advantageous to fabricate metamaterials with AM. Many common AM techniques, such as SLM, SLS, EBM, DMLS, FDM, and SLA, have already been used to successfully fabricate metamaterials. The drawback to using these methods are related mostly to precision of the build, which is generally at the upper range (sub-millimeter) of what is desired in metamaterial fabrication. To address this, a number of high precision AM methods have been developed that can achieve submicron and even nanometer scale resolutions. The most prominent of these methods is projection microstereolithography, which is based on SLA, able to achieve micron level precision, scales well to larger parts, and is a relatively simple and well understood process. The high precision AM methods developed so far have primarily been intended for non-metallic materials but some recent work has started to modify the processes to print metallic and ceramic materials. Metallic and ceramic metamaterials have the ability to offer structural stiffness in addition to the other desired properties and are an area that need further development, especially the development of high precision metallic AM methods.

The primary purpose of AM in metamaterial fabrication has not necessarily been to induce novel properties, but rather AM has functioned as an enabling technology for metamaterials. In theory, any metamaterial property can be realized without the use of AM but in practice, without AM, many metamaterials would never be able to demonstrate the theoretical properties desired. With the increasing popularity of computational simulation and topology optimization, metamaterials with extremely complex structures can be designed to exhibit more than just a single behavior, such as cloaking or negative refractive index. The metamaterial can be designed so that it demonstrates the desired property while also maintaining optimal cooling, flow, or strength for instance. Since AM is inherently a computationally driven process, it is realistically the only fabrication method available to fabricate such designs. Mechanical metamaterials are generally based on periodic lattice structures, which could be somewhat easily constructed with non-AM fabrication methods. However, the use of AM opens the design space so that disordered and/or multimaterial lattices can be fabricated. Multimaterial, mechanical metamaterials have demonstrated Poisson ratio's as low as -7, much lower than single material metamaterials have shown. In acoustic metamaterials, axisymmetric acoustic cloaks and radar absorbing metamaterials have been created using designs that can only, practically, be fabricated with the use of AM. The use of AM to construct electromagnetic, optical, and thermal metamaterials has not been widely examined, likely due to the sub-micron resolutions required in those metamaterial structure. With high precision AM methods quickly being developed, these metamaterial classes are ripe with opportunity for new developments. AM of ceramics is increasing in popularity and could enable designs of thermal metamaterials at extreme temperatures that serve as thermal cloaks for jet engines, as an example.

Another area of development in metamaterials is in 4D printing (AM of smart materials) applications. The preliminary work in this area has shown tunable and switchable metamaterials. The additional degree of freedom of time in the design space allows for metamaterial designs that can modify their behavior based on an environmental stimulus. The develop of 4D metamaterials is very new and has generally only been demonstrated in a laboratory setting and mostly with polymeric materials. The usage of metallic smart materials, like shape memory alloys (SMAs), offers even more possibilities, especially when combined with shape memory polymers (SMPs) to create "freezable" shape memory composites. Multimaterial AM methods could easily enable such materials to be fabricated. One possible design could utilize AM to embed SMAs in an SMP-based origami metamaterial. This metamaterial selection, and could find application in space-based deployable structures or biomedical applications.

The use of AM to fabricate metamaterials is in many ways limited by the AM technology and not necessarily an understanding of the metamaterial. However, using AM to fabricate metamaterial makes new designs of internal metamaterial structures possible and advances the state of the art in metamaterials. The unique structure of metamaterials mean that what is best for general AM technology may not be best for metamaterial development and purposeful research should be conducted to maximize the benefits of AM for metamaterials. As one example, porosity and anisotropy in AM is generally thought to be a negative feature but in metamaterials, these defects could be used to tailor mechanical properties of the metamaterial.

METAMATERIALS USING ADDITIVE MANUFACTURING TECHNOLOGIES

1. INTRODUCTION

The topics of both metamaterials and additive manufacturing (AM) have no shortage of existing literature. As of the submission of this work, searching the terms "metamaterials" and "additive manufacturing" in the Scopus database yields over 35,000 and over 24,000 publication results, respectively. Plotting these results by publication year (Fig. 1), shows the rapid rise in popularity of both technologies in the past 20 years. While both AM and metamaterials are popular fields, the overlap of the two technologies (*i.e.* using additive manufacturing to produce metamaterials) is just starting to emerge.



Fig. 1—Scopus publication results per year for the search terms "metamaterials" and "additive manufacturing".

In this report, a summary of recent literature on the use of AM to fabricate metamaterials will be shown. The remainder of this report will be structured as follows: Section 2 briefly introduces the background of AM as well as metamaterials; Section 3 examines the fabrication of metamaterials and AM processes that can be applied to fabricate metamaterials; Section 4 reviews recent work on metamaterial parts that have been fabricated using AM and the unique properties and applications of those parts; finally Section 5 identifies potential opportunities for fabrication and application of AM metamaterials.

2. ADDITIVE MANUFACTURING AND METAMATERIALS

As mentioned, both AM and metamaterials are fields with a vast body of literature. Both AM and metamaterials encompass a very broad range of research and applications. In this chapter, a brief overview of AM and metamaterials is presented.

Manuscript approved March 29, 2020.

2.1 Additive Manufacturing Technologies

Much of the work in additive manufacturing (AM) has been accomplished in the past two decades but the ideas used in AM can be dated to as far back as the 1860s. [1] AM is referred to by many names in the literature including, but not limited to, 3D printing, rapid prototyping, and layer manufacturing. [1– 4] The general AM process has, however, been specifically defined by ISO/ASTM standards to be the "process of joining materials to make parts from 3D model data, usually layer upon layer, as opposed to subtractive manufacturing and formative manufacturing methodologies", which gives the area of AM a very broad scope. Likewise, by the same definition standards, major AM processes can be generalized into seven categories: fused deposition modeling (FDM), inkjet printing (IJP), laminated object manufacturing (LOM), laser engineered net shaping (LENS), stereolithography (SLA), selective laser sintering (SLS), and threedimensional printing (3DP). [2, 5] As with the naming convention for AM, the naming of the processes can vary widely [1], where, for example, technologies that all use laser processing for metallic materials can be referred to as different names depending on how the powder is fed and melted. [6-8] AM can utilize a vast range of materials from plastics to metals to ceramic to foods and other biological/polymeric materials. [4, 9– 12] Typically, certain classes of materials are printed with certain method (e.g. metals with SLS or LENS, plastics with FDM, etc.) but this is not a strict guideline and as reserach in the area expands, materials able to be printed using a given AM method will also expand. [11, 13] A summary of the classification of AM processes can be seen in Figure 2. [14]

CATEGORIES	TECHNOLOGIES	PRINTED "INK"	POWER SOURCE	STRENGTHS / DOWNSIDES
Material Extrusion	Fused Deposition Modeling (FDM)	Thermoplastics, Ceramic slurries,	Thermal Energy	Inexpensive extrusion machine Multi-material printing Limited part resolution Poor surface finish
	Contour Crafting	Metal pastes		
Powder Bed Fusion	Selective Laser Sintering (SLS)	Polyamides /Polymer	High-powered Laser Beam	 High Accuracy and Details Fully dense parts High specific strength & stiffness Powder handling & recycling Support and anchor structure Fully dense parts High specific strength and stiffness
	Direct Metal Laser Sintering (DMLS)	Atomized metal powder (17-4 PH		
	Selective Laser Melting (SLM)	stainless steel, cobalt chromium, titanium Ti6Al-		
	Electron Beam Melting (EBM)	4V), ceramic powder	Electron Beam	
Vat Photopolymerization	Stereolithography (SLA)	Photopolymer, Ceramics (alumina, zirconia, PZT)	Ultraviolet Laser	 High building speed Good part resolution Overcuring, scanned line shape High cost for supplies and materials
Material Jetting	Polyjet / Inkjet Printing	Photopolymer, Wax	Thermal Energy / Photocuring	Multi-material printing High surface finish Low-strength material
Binder Jetting	Indirect Inkjet Printing (Binder 3DP)	Polymer Powder (Plaster, Resin), Ceramic powder, Metal powder	Thermal Energy	 Full-color objects printing Require infiltration during post- processing Wide material selection High porosites on finished parts
Sheet Lamination	Laminated Object Manufacturing (LOM)	Plastic Film, Metallic Sheet, Ceramic Tape	Laser Beam	 High surface finish Low material, machine, process cost Decubing issues
Directed Energy Deposition	Laser Engineered Net Shaping (LENS) Electronic Beam Welding (EBW)	Molten metal powder	Laser Beam	 Repair of damaged / worn parts Functionally graded material printing Require post-processing machine

Fig. 2-Classification of AM processes by ASTM International. [14]

The adaption and expansion of AM can in part be attributed to the reducing cost of programmable controllers, lasers, and computer-aided design (CAD) software. [14] The process of creating a part to be built by AM begins with 3D solid modeling as a CAD file which is then sliced into many discrete layers

that are used as inputs to the AM machine to build the part. The largely digital workflow of the AM process allows for significant design and geometric flexibility with high dimensional accuracy. The AM process is also cost efficient as there is no need for assembly of parts and there is very little waste material. [13, 14] Additionally, AM processes can be scaled to print parts that range from the nano/micrometer scale [15–17] to the scale of meters. [18, 19]

2.1.1 4D Printing

4D printing is a subset of 3D printing (or AM) that is applied when smart materials or multi-material systems are used in the AM process with the capability to transform over time. [20, 21] The transformation of the material system in time is a result of an applied stimulus in the form of thermal, mechanical, electrical, magnetic, chemical, and/or many other forms or combinations of stimuli. [22–24] The bulk of research in 4D printing is centered around shape memory alloys [25], polymers [26, 27] and their composites [28] but recent advances have begun to extend this to multi-material systems and metamaterials. [29] Applications of 4D printing are growing as well with 4D parts being built for applications in biomedicine, textiles, soft robotics, oragami, adaptive joints/structures, and aerospace. [24, 27]

2.2 Metamaterials

As with AM, metamaterials have exploded in popularity in the past 20 years but the origins of these materials can be traced back to the early nineteenth century. [30] Metamaterials are fabricated structures and composite materials that either mimic known material responses or qualitatively have new, physically realizable response functions that do not occur or may not be readily available in nature. [31] These "artificial" materials can have exotic properties, which are not primarily dependent on the intrinsic properties of the chemical constituents, but rather on the internal, specific structures of metamaterials. [32] The origins of modern metamaterials are derived from the theoretical work of Veselago [33] then, later, Pendry [34] and Smith *et al.* [35]. However, these works were purely theoretical and it was the work of Shelby *et al.* [36] that first fabricated such a material and demonstrated a negative refraction index.

The underlying scheme of metamaterials is that the structures manipulate wave physics to achieve unusual or previously unachievable behaviors. [37] As mentioned previously, the earliest works in metamaterials were in electromagnetics to realize the theoretical double negative (DNG) material (Fig. 3a) of Veselago's early work. The work in artificial magnetism of Pendry *et al.* in the 1990's was the first to demonstrate a magnetic material with negative permeability (μ) and positive permittivity (ε). [38] This work was key to the development of an artifical structure known as a split-ring resonator (SRR), which first demonstrated a realizable DNG. [34, 36, 39] Initially, these metamaterials were limited to microwave ranges (Fig. 3b) [40] but research quickly expanded the area to encompass frequencies in the terahertz and optical range. [41, 42] The structures originally proposed and built to create and demonstrate these metamaterials were single functional layers but Valentine *et al.* [43] realized a 3D structure by appropriately stacking several individual layers. [44, 45] The success of metamaterials in electromagnetics/optics led to a natural extension of metamaterials that were designed to operate at much larger wavelengths and length scales such as those in acoustics, thermodynamics, and mechanics. [46, 47]

2.2.1 Thermal Metamaterials

Thermal metamaterials are governed by equations similar to those used in electromagnetic metamaterials [48] and are referred to as transformation thermodynamics. [49, 50] Thermal metamaterials are perhaps the



(a) Classification of metamaterials by permittivity (b) Recent progress in electromagnetic/optical and permeability. [31] DPS: double positive, DNG: metamaterials. [45] double negative, MNG: μ negative, ENG: ε negative.

Fig. 3—Classification and progress in electromagnetic/optical metamaterials.

least studied class of metamaterials due to the relatively late derivation of transformation thermodynamics. [51] However, thermal metamaterials are quickly being advanced for applications in thermal cloaking, protection, and management. [52, 53]

2.2.2 Acoustic Metamaterials

Acoustic and mechanical metamaterials are similar and in many cases grouped together since acoustics depend on the bulk modulus in a similar manner as classical continuum mechanics. [54] The first acoustic metamaterials (AMMs) developed by Liu *et al.* [55] used rubber-coated spheres to create locally resonant and deeply subwavelength structures that responded to incident acoustic waves. Since then many reserachers have investigated AMMs for their ability to control the attenuation of acoustic waves, invisibility cloaking capabilities, and acoustic wavefront engineering, such as focusing via manipulating the acoustic impedance of metamaterials. [56] AMMs are able to control wave propagation by changing the bulk modulus (κ) and/or the mass density (ρ) in much the same way that is done with electromagnetic metamaterials (Fig. 4). As such, it can be shown that κ^{-1} corresponds to ε and ρ to μ in the acoustic and electromagnetic governing equations. [47] The literature is vast in the area of AMMs and there are a number of very thorough reviews on the topic of AMMs which discuss acoustic hyperlens' [57], mechanisms of absorption [58], tunable AMMs [56], and other aspects and applications such as cloaking and active AMMs. [59–61].

2.2.3 Mechanical Metamaterials

General mechanical metamaterials are perhaps the most complex metamaterials to design due to the number of possible parameters involved in describing a continuum material (*i.e.* anisotropic elasticity tensor, thermo-elasticity, elasto-plasticity, *etc.*). [48] Mechanical metamaterials are typically engineered to yield properties such as zero or negative Poisson ratio (auxetic) [62–65], very low mass or negative mass density (light-weight), vanishing shear modulus (pentamode or anti-auxetic) [66, 67], negative stiffness or compressibility (negative parameter) [68, 69], origami structures, and/or negative coefficient of thermal expansion [70, 71]. Examples of each of these classes of materials can be seen in Figure 5.



Fig. 4—Classification of acoustic metamaterials by mass density, ρ , and bulk modulus, κ , in the same manner as Figure 3a. [60]

While most mechanical metamaterials are metallic, research in the area of polymers [72] and ceramics [73] have also been studied. As with AMMs, research in the area of mechanical metamaterials is vast and for further information, the reader is directed to some recent reviews of the field. [54, 74–76]

3. FABRICATION AND ADDITIVE MANUFACTURING OF METAMATERIAL

As has been discussed in the previous section, metamaterials behavior is dependent on the internal structure of the metamaterial. This structure is in general, a complex, small repeating structures. As one can imagine this structure is extremely suitable to be manufactured by AM technologies due to the many unique advantages it offers, such as the ability to print geometries from the macro to micro scale and being free from space limitations. [77, 78] This chapter will address general metamaterial fabrication briefly then review recent work on fabrication of metamaterials with AM. Specific properties and applications of AM fabricated metamaterials will be discussed in Section 4.

3.1 Metamaterial Fabrication

Since the first metamaterials were demonstrated in 2001, enhanced fabrication techniques for metamaterials in both 2D and 3D have been of interest. The single, largest, current barrier to metamaterial widespread adaption is the difficulty in s caling from 2D to 3D, and doing s o in a manner that maintains sufficient resolution of the desired unit cell while fabricating parts at the macro scale.

The most common class of methods for 2D metamaterial fabrication is lithography. Lithography for metamaterials generally takes the form of photolithography, shadow mask lithography, soft lithography, electron beam lithography, x-ray lithography, or two photon lithography. [79] Photolithography was among the earliest method of fabricating metamaterials. Generally speaking, lithography methods use a stamping or transfer printing process requiring a "negative" of the design to be printed and are exposed to an energy source such as UV light or an electron beam. These methods can produce features at the sub-micron and nano scales, and can be scaled to 3D by stacking multiple layers. [80] However, lithography methods are typically slow, expensive, and lack scalability to large areas. [81] Stereolithography is a method commonly used in AM but is not widely used for metamaterial fabrication due to insufficient resolution. However,



Fig. 5—Overview of mechanical metamaterials by Christensen *et al.* [54]. B: bulk modulus, ρ : mass density, and G: shear modulus. In the first column, red represents the metamaterial regime while black represents traditional materials.

recent advances in microstereolithography have resolved this issue. [82] Further discussion on this topic is delayed to the next section.

Ideally, large-scale 3D metamaterials will be achieved through some low cost, high throughput process but, at present, most fabrication methods fall into either 3D or large-scale with some techniques beginning to bridge the gap. One of the first 3D fabricated metamaterials was by Deubel *et a l.* [83] using direct laser writing, but this method suffers from low throughput and material selection issues. [80] Other advanced 3D fabrication techniques such as imprint lithography, interference lithography, multi-photon polymerization [17], and focused ion beam milling have been studied but generally suffer from the same issues of low throughput, lack of resolution, lack of scalability beyond the micron scale, and/or limited material selection. [79, 84]

Additional 3D methods derived from traditional manufacturing processes such as ceramic casting [85] and injection molding [86] have also been adapted to metamaterial fabrication but these have not been widely adapted due to issues with mold creation and required sintering for final part f ormation. The most promising methods for fabricating large-scale 3D metamaterials are nanoimprint lithography [87], pattern transfer [88], self-assembly [89, 90], and AM. Deatils of all but AM are omitted here but the reader is referred to Yoon *et al.* [80] and Bishop-Moser *et al.* [81] for an overview of these methods.

3.2 Additive Manufacturing of Metamaterials

AM has been used to build light-weight, high stiffness structures through topology optimization since its inception and its use has only become more widespread. [14, 91] While many examples of AM-built topology optimization are truly freeform structures, many are based on patterned unit cells. [92] Thus, it is natural to pair AM and metamaterials. However, most common AM techniques operate at the micron to meter scale while metamaterials typically use structures in the millimeter to nanometer scale, especially in electromagnetic and optical applications where the substructure of the metamaterial needs to be of the same size as the wave length. In order to accommodate this scale difference, many new high precision AM techniques have been developed/implemented for metamaterial fabrication.

There are a number of works that use a modified high precision version of stereolithography (SLA) called projection microstereolithography (P μ SLA). [78] This method is capable of producing features with 1 μ m resolution in a nearly identical fashion as traditional SLA. P μ SLA is perhaps one of the most common metamaterial fabrication techniques used in the literature. [82, 93, 94] This is in part due to the fact that it is a relatively simple process that is well understood, has a throughput, and has been scaled well to large build planes. As with traditional SLA, P μ SLA is typically limited to photopolymers but some work in ceramics [95] and metals [96] has been conducted.

Two other high precision methods of metamaterial fabrication by AM, which are both somewhat common, are direct laser writing and direct ink writing. The former shares some similarities with SLA while the latter more closely resembles binder jetting or material extrusion methods. [97] Within direct laser writing is a method known as two-photon polymerization (TPP), which uses a femtosecond-pulsed laser beam to initiate the photopolymerization process. This method is extremely accurate with resolutions under 100 nm and can operate in a truly 3D space (*i.e.* no support structure is required and photopolymerization happens on a continuous 3D laser path rather than a layer by layer approach). [98] TPP has a very small material selection and while extremely precise, does not scale to large parts well. In direct ink writing, colloidal inks are used which solidify based on pressure and not phase change as traditional extrusion methods. [99, 100] Direct ink writing can reach sub-micron resolutions and has a wide selection of materials but suffers from slow printing processes. [78, 101]

Before continuing, two other methods of high precision metamaterial fabrication are noted. Electrophoretic deposition [102] and electrohydrodynamic printing [103] both use electrostatic fields to transport material to the desired location. Both processes are very promising as they are low cost, reliable, high precision methods. However, the processes suffer from a number of drawbacks including low rate deposition. [78] Additionally, while both methods are additive, neither fits in the standard classification of AM processes (Fig. 2) and as such will not be discussed further.

While the techniques discussed above are high precision AM methods, there are still a number of applications in metamaterial fabrication that make use of traditional AM methods, particularly in mechanical

and acoustic metamaterials where metamaterial structures need not be micro/nanoscale. To avoid repetition, only a brief overview of some traditional AM methods that have been used for metamaterial fabrication are given.

SLA has been used to produce mechanical octet lattices for energy absorption [104] and metamaterial embedded geometric optics [105]. Powder bed fusion methods such as SLS [106–109], DMLS [110], SLM [15], EBM [111] have all been extensively implemented, particularly for mechanical metamaterials where the strength of metallic structures is desired. FDM has also been used in 4D metamaterials [112] and other polymer applications [113]. Other methods such as binder jetting [114] have been utilized but their usage is sparse. While not directly related to directed energy deposition (DED), aerosol jet printing [115] shows some similarities DED as well as aerosol deposition (AD) [116] and cold spraying (CS) [117] but with much more precision.

Many of the AM processes reported in Figure 2 are seen in the above overview. Methods such as sheet lamination and DED seem to be the least common methods of fabricating metamaterials. Sheet lamination is most likely not used since it does not have a mechanism to produce the structural features required for metamaterials. DED generally does has a lower resolution than other comparable sintering or melting methods for metallic materials, which likely explains it's lower uptake in the metamaterial fabrication process. However, DED has the ability to operate with multi-axis milling machines thus has more freedom spatially than other AM processes. This freedom could be beneficial in c reating c omplex metamaterial structures and doing so may aid in the creation of isotropic metamaterials, a highly desirable feature. DED also has the ability to create functionally graded and multi-material parts, which other methods, especially metallic AM methods, cannot do.

3.2.1 4D Metamaterials

As previously mentioned, another emerging class of 4D materials are 4D metamaterials. Due to the relatively recent uptake of 4D printing, work in 4D metamaterials is limited but provides the prospect of creating materials that function as multi-state metamaterials. [118] The most common methods of 4D metamaterial printing are FDM and SLA/P μ SLA for shape memory polymers (SMPs) but other methods such as SLS and inkjet pritning for SMP fabrication and SLM for shape memory alloys (SMAs) have been successful. [24, 119–121] Other works in 4D printing have started to examine functionally graded printing and multi-material printing but these techniques have not been used to demonstrate conventional metamaterial behaviors. [122]

4. ADDITIVELY MANUFACTURED METAMATERIALS

In this section, properties achieved by AM fabricated metamaterials and some applications will be discussed. As one would expect, AM fabricated metamaterials exhibit many of the same properties and non-AM metamaterials but effort to discern the two classes will be made. It should also be mentioned that a recent review by Wu *et al.* [123] provides an excellent overview of additively manufactured metamaterials and their applications/properties. The current work differentiates itself by giving a broader overview of applications and properties rather than many examples of prominent metamaterial usages.

4.1 Properties

As in Section 2.2, electromagnetic and optical metamaterial properties are discussed first, followed by thermal, then acoustic, and finally m echanical. Few authors purely discuss properties of electromagnetic or optical metamaterials that have been fabricate by AM. Rather, most authors take lesson learned from electromagnetic materials and design a metamaterial for a specific application then u se A M to p rint the part. Intuitively, this makes sense as electromagnetic metamaterials have been extensively studied in 2D and their properties are generally well understood so the use of AM to fabricate the part is purely an application driven method. Additionally, the small scales required for electromagnetic metamaterials make AM of even simple structures a challenge. However, some authors do discuss properties that can only be achieved by AM. For instance, Garcia [124] describes the use of FDM to build a part based on unit cells that yielded an anisotropic dielectric tensor. Along the same lines, Isakov *et al.* [113] used high and low dielectric permittivity materials, layered appropriately, to achieve artificial anisotropic dielectric permittivity.

4.1.1 Thermal Metamaterials

As with AM electromagnetic materials, there is only a small body of literature in the design of tailored properties for thermal metamaterials by AM without a specific application in m ind. Wu *et a l.* [125] used multimaterial inkjet printing to design and fabricate antichiral structures that exhibited a negative thermal expansion. Similarly, Wang *et al.* [70] used multimaterial SLA to design a lattice structure that had a negative thermal expansion. Specific work on thermal cloaking, negative thermal conductivity, *etc.* that was enabled by AM does not appear to be an active area of research and thus has many opportunities for new studies such as designing metamaterial substructures that could only be produced by AM.

4.1.2 Acoustic Metamaterials

Locally resonant metamaterials are designed to prevent waves from propagating in certain frequency regions (bandgaps) for acoustic insulation or cloaking purposes. In 2013, Sanchis et al. [126] used AM to develop a prototype of an axisymmetric cloak, which was able to suppress 90% of the waves scattered by a sphere at a certain frequency. In acoustic cloaking, it is particularly important for a cloak to be isotropic and omni-directional. The techniques demonstrated by Sanchis et al. could be extended in such a way that an omni-directional cloak could be designed and enabled by AM. Raza [127] later made locally resonant metamaterials for energy absorption similar to Sanchis et al., but Raza used a novel multi-material printing method to fabricate the part. The method used was a hybrid droplet direct write printing and continuous direct writing printing method with 3 different materials, 2 polymers and an adhesive. This technology could be significant if slightly modified to expand the range of materials us ed. For in stance, Matlack et al. [128] designed a composite structure of polycarbonate and steel cubes for vibration absorption. The polycarbonate used in the study was 3D printed but paused intermittently to place the steel cubes. If the technology developed by Raza were modified properly, it is conceivable that a complete composite structure could be made without interruption. In another study, Wormser et al. [111] used EBM AM with titanium to design an LRM based on an optimized 3D lattice. Multiple designs were found, fabricated, then tested and showed different phononic band gaps and a slightly auxetic behavior. The dual purpose nature of the designed structure leads one to the conclusion that it may be possible to design metamaterials with both structural and acoustic or thermal purposes.

4.1.3 Mechanical Metamaterials

In comparison to other metamaterials, mechanical metamaterials produced by AM have a large body of literature. This can most likely be contributed to the fact that the AM process is a thermomechanical and materials driven process, which has obvious strong ties to structural mechanics and therefor, mechanical metamaterial design. Additionally as mentioned earlier, mechanical metamaterials do not require the same resolution as electromagnetic and optical metamaterials thus making them easier to fabricate and more readily available to study.

One of the earlier uses of AM methods to build mechanical metamaterials was Kashdan et al. [106], where a negative stiffness element was fabricated out of Nylon by SLS. The element fabricated was intended for vibration isolation and was a prototype which used a spring to control the buckling behavior of a beam. This prototype was a unique design for a metamaterial but does share some similar design elements as the negative-parameter design in Figure 5. SLS has been used in other works to construct disordered lattices for mechanical constant tuning [109] and fabrication of an auxetic structure, which used polyamide impregnated with carbon nanotubes (CNTs) [108]. The disordered lattices created by Mirzaali et al. [109] were unique since most metamaterial substructures are based on symmetric octet, re-entrant, or similar periodic lattices. Since the disordered lattices were complex and AM was required in order to fabricate them. These disordered structures were able to achieve a much broader range of elastic modulus-Poisson ratio duos than typical lattice restricted networks. AM was also a necessity to fabricate the CNT-modified polyamide auxetic bucklicrystals of Yuan et al. [108]. This work was able to show how energy absorption characteristics could be modified as a function of the design of the structures. Additionally, the use of nonmetallic materials modified by CNTs were able to avoid processing defects inherent in most metallic AM. The use of CNTs in this work could be extended to give the auxetic structure more unique electrical/thermal properties that are commonly displayed by CNT-based materials.

Other AM methods, such as EBM, SLM, and SLA, have also been used to fabricate mechanical metamaterials. Amendola *et al.* [129] used EBM to fabricate a variety of titanium based pentamode metamaterial structures to study how the micro- and macroscopic features affect the structure, intended for vibration isolation applications. Some cyclic tests were completed but it would be interesting to examine if long term fatigue under low vibration loads is significantly effected by the metamaterial structure. Similar to the previous work, Hedayati *et al.* [130] used SLM to fabricate titanium pentamode materials. The novelty of this work was the variable energy density used to produce different strut shapes in the design of a pentamode metamaterial. In another study, Tancogne-Dejean *et al.* [15] used SLM to manufacture 316L stainless steel pentamode metamaterials where the strut dimensions were varied to produce different behaviors in the macroscopic structure. Additionally, this work showed results of the actual metallic microstructure where defects in manufacturing were analyzed. These defects are well known in the study of the AM process and more work in this area for metamaterials could lead to tailored microstructures that further enhance mechanical properties or possibilities beyond what is currently being studied.

Pentamode metamaterials have also been studied with polymeric materials printed by SLA. The printed lattice structures in the metamaterials have been varied [131] and shown high energy absorption characteristics with large recoverable strains [104]. These phenomena have also been shown at smaller scales from pentamode metamaterials produced by P μ SLA. [82, 94] The metamaterials have shown very light weight structures with near linear mass-stiffness scaling in contrast to standard materials, which follow a square or cube relationship. P μ SLA has also been used to print multimaterial lattices with tunable mechanical constants including Poisson ratio's that can range from nearly 0 to -7. [132] The range and control of elastic

constants in the demonstrated multimaterial lattices show the possible range of the multimaterial design space enabled by AM. Combined with topology optimization and computational simulations, this design space could be exploited to architect metamaterials with behaviors even more extreme than those seen in the previous work. In addition to $P\mu$ SLA, direct ink writing has been used to produce structures exhibiting the same near linear mass-stiffness scaling and negative stiffness. [101]

Another emerging field in mechanical metamaterials is heirarchical structural design. Hierarchical design is enabled by the ability to create structures at the nano/microscale (by using such methods as self assembly, TPP, or P μ SLA) and incorporate those features into much large macroscale structures. [107] Among the first groups to study this type of structure was Meza, Zheng *et al.* [96, 133, 134]. Starting at the nanolattice, the group was able to use large area P μ SLA to construct nanoscale substructures that could take advantage of the material size effect and propagate this all the way up to a part on the centimeter scale. Parts constructed at using this hierarchical approach could yield extreme mechanical properties, tailored to any desired application, while maintaining extremely low densities. Additional, due to the scale of the base nanolattice, it may be possible to use this technique to construct metamaterials other than mechanical metamaterials, such as optical and electromagnetic or thermal metamaterials.

4.1.4 4D Properties

As discussed previously, 4D printing enables unique metamaterial properties such as materials that act as switches or can freeze in specified positions. [112, 120] This type of metamaterial could be designed as an auxetic metammaterial and has the potential be used in confined applications such as deployable space structures. The metamaterial could be stored in a compacted state then released and expanded to a much larger form using only environmental stimulii. Additionally, the transforming nature of the material could enable tunable band gaps that can interact with an entire range of frequencies rather than a single frequency as demonstrated by previous works. [135] The majority of 4D metamaterial printing has been conducted on SMPs, but SMA are an emergening area. [136] These materials have not been specifically constructed as mechanical metamaterials so far, but the lesson learned from the current work can be easily extended to metamaterial design. For instance, Saedi *et al.* [137] showed that porous NiTi could be manufactured by SLM and based on the porosity, the mechanical properties could be tuned. Detailed simulation and optimization could extend the ideas demonstrated by Saedi *et al.* to construct mechanical, thermal, and/or acoustic metamaterials which are tunable depending on the application.

4.2 Applications

Additive manufacturing has not only enables unique properties in metamaterials, unattainable by conventional processing methods, but has also enabled the construction of parts for unique applications. The multimaterial system developed by Raza [127] was used to fabricate different LRM designs, which were used for energy harvesting. Likewise, Yin *et al.* [138] developed and demonstrated a radar absorbing structure fabricated by SLA. This application used alternating "woodpile" unit structures and patterned them in such a way that they become more dense near the structure to be protected from radar. This type of design can only be enabled by the freeform fabrication abilities of AM.

As previously mentioned, Sadeqi *et al.* [105] used SLA to partially develop a metamaterial embedded geometric optic. Through the use of AM, Sadeqi *et al.* were able to cover an optic in microwave absorbing metamaterials which absorbed a designed frequency. The authors noted that with increasing resolution and area, THz and optical metamaterials could be embedded in the optic. It is postulated that a system such

as the one developed by Meza, Zheng *et al.* [96, 133, 134] combined with SMP printing could enable high resolution printing over the large desired area and also give a frequency tunable optic from the SMP actuation.

A class of mechanical metamaterials not yet discussed are those involved in origami structures as shown in Figure 5. This class of metamaterial uses periodic arrangements of folds to generate design with high flexibility, potential for deployability, and low w eight. A number of applications are discussed in Zadpoor [74] but one of the most interesting of these are the use of 4D printing to construct the origami. This enables self folding and unfolding of the origami, greatly increasing its applicability. Liu et al. [139] used FDM to create 4D Miura-ori origami using SMPs, which showed 300% reversible volume change through only a thermal stimulus. Han et al. [140] have recently developed a multimaterial P μ SLA system and suggest that it could be used for smart material printing. Han et al. used this machine to demonstrate a multi-responsive hydrogel beam through thermal and electrical inputs. Active origami is a field which uses SMAs embedded in elastomers to generate self-folding origami structures. [141] In this field, shape memory composites using SMA wires embedded in SMPs has been suggested as a mechanism to create self-folding origami that is able to stay locked into its actuated, folded position. [142] The utilization of a system like the one developed by Han et al. could expanded the possible design space of shape memory composite origami by enabling the simultaneous printing of SMAs and SMPs. The SMAs would not necessarily have to be embedded as wires but could be printed as a planar grid within an SMP that was printed similarly to the one printed by Liu et al.

The discussed applications are in no way exhaustive and AM can be used to fabricate almost any metamaterial for any application. There are, of course, restrictions such as part size to be built and material selection but these types of hurdles would need to be overcome with traditional fabrication as well. Some additional applications not discussed can be found in review articles of metamaterials and AM usages such as [13, 78, 97]. Some applications for AM fabricated metamaterials suggested in these works are supercapacitors, battery electrodes, and biomedical applications, which will be discussed briefly in the coming section.

4.2.1 Biological Applications

AM use for biological/biomedical applications is an innovation that makes use of the relatively low throughput of AM and the flexibility in design to create patient-specific structures and medical phantoms. These type of personalized parts could be for anything from dentures to hearing aids to prosthetics and recent work has begun to make use of metamaterials in biomedical applications. [143] For instance, a metamaterial can be fabricated using AM to exhibit auxetic behavior and this can be used to control local cell density for cell growth/regeneration.

Biological tissue is among the most mechanically complex materials to characterize and can exhibit hyperelasticity, viscoelasticity, incompressibility, and anisotropy with some materials exhibiting all these behaviors simultaneously. Additionally, biological materials (*e.g.* bone) can contain hierarchical structures or complex microstructures such as those seen in trabecular or cancellous bone. These features make metamaterials, which are based on lattice structures and can exhibit a near infinite range of mechanical properties, ideal for bio-mimicking materials. Johnson *et al.* [144] have done exactly that and used AM to fabricate metamaterials which were able to mimic a range of tissue behaviors. The use of metamaterials allowed them to use well understood elastic bio-compatible materials to create the lattices needed to mimic the bio-material and AM made the lattice construction possible. Similarly, Wang *et al.* [145–147] examined

multimaterial AM to create tissue-mimicking phantoms. This multimaterial approach even more closely approximates the tissues and with further advances, it is conceivable that AM could be one day used to create entire limbs, which would aid in understanding of injurious events. In another work, EBM and titanium was used to fabricate auxetic metamaterials combined with traditional materials to create "meta-implants", which were able to reduce the risk of bone-implant interface failure. [148] The use of both metamaterials and conventional materials combined with the patient specific nature of the product could only be enabled by AM. Beyond just tissue-mimicking phantoms that capture the biomechanical response, AM fabricated metamaterials are able to grow/regenerate tissue using implanted cells. Raman *et al.* [93] used multimaterial $P\mu$ SLA to create a cell and biomaterial pattern which replicated neurovasculature. The use of AM here was crucial to create the proper patterning which most closely mimicked live tissue so that the cells would be as close to *in-vivo* as possible. It is suggested that this type of multimaterial AM could be used in a wide range of applications, beyond the neurovasculature, including high-throughput drug testing and *in-vitro* disease development studies.

5. SUMMARY AND FUTURE WORK

This work has presented an overview of additively manufactured metamaterials including their unique properties and applications. A brief overview of AM and metamaterials was shown followed by a review of metamaterial fabrication methods. It was shown that many common methods used to fabricate metamaterials are similar to those used in AM and that the extension of metamaterial fabrication to AM was a natural and inevitable one. A review of AM process currently used for metamaterials fabrication was then shown and found that many AM technologies have already been used to fabricate metamaterials. However, many of these processes needed to be modified in order to be applicable to the small (nano/micro) scale required in many metamaterial designs, such as those in electromagnetic and optical applications. In the previous chapter, unique properties and applications of metamaterials fabricated by AM technologies were shown. Throughout the report, a number of knowledge gaps and opportunities for future developments have been made, a few of these points will be elaborated on now.

AM Fabrication Method Opportunity — As discussed, directed energy deposition (DED) is not an AM method that has been used for fabrication of metamaterials. The abilities of DED methods should be explored further since they can be applied to metallic materials and additionally have the capability to produce multimaterial and functionally graded parts, which powder bed based method cannot do. It is postulated that the reason for the lack of adoption in metamaterial fabrication is due to the relatively low dimensional accuracy in the method. However, the DED process is comparable to the aerosol jet printing method of Jahn et al. [115] with the exception that DED uses a laser to sinter the material as it is sprayed. High precision lasers are available such as those used in two-photon polymerization and combined with aerosol jet, could in essence result in a high precision DED method. Such a development could generate possibilities for unique metallic metamaterials with thermal, acoustic, or electromagnetic/optical properties in addition to high structural strength. This type of development could also be combined with functional grading of the material and topology optimization to yield structures that maintain their strength, decrease density, and could still function as metamaterial cloaks. Another, more easily accessible avenue for DED implementation to fabricate metamaterials is the use of a hybrid-AM system. This system utilizes both additive and subtractive techniques to produce high quality, high resolution parts. CNC subtractive processes can be as precise as a single micron, which could provide sufficient a ccuracy f or m any metamaterial applications.

AM Technique Development — Another possible development in AM technologies that could enhance the current state of art metamaterial fabrication is multi-resolution laser processing. In this theorized system, there would essentially be a macroscale laser that would shape the overall part and fabricate a large portion of the volume, then there would a more focused laser that would sinter the fine details in the metamaterial structure. Alternatively, the system could use a single laser that is able to be focused to different scales on the fly. The envisioned system would likely be implemented in a powder bed method for metallic materials. A major hurdle of this process would be powder processing and control as the powder particle would need to be fine enough to produce the required detail in either scenario. A system such as this, that could produce high resolution metallic parts, could aid in expansion of electromagnetic, optical, and thermal metamaterials fabricated by AM, thus giving researchers/engineers more design freedom while being able to maintain structural rigidity.

Microstructure Refinement and Mechanical Properties — In printing metallic materials with AM, it is well known that the microstructure is unique and very far from traditionally manufactured parts of the same material, typically exhibiting anisotropy and heterogeneity (*i.e.* porosity). It is also known that this unique microstructure is dependent on the AM process parameters and affects the mechanical behavior of the printed part greatly. In addition to the microstructure, residual stresses from the thermal processing are commonly developed and can warp finished parts, especially finer structures. The understanding of how residual stress and the unique microstructure are developed, and how they effect the final mechanical properties is highly studied in the AM community (see process-structure-property, PSP, linkages in the literature). This type of work needs futher development in the area of AM fabricated metamaterials, especially those which require high resolution structures that are particularly susceptible to warping. The understanding of residual stresses and microstructure anisotropy and heterogeneity could lead to potential developments where the microstructure could be tailored to further enhance the metamaterial properties, e.g. utilizing/manipulating the anisotropic stiffness tensor throughout the structure to achieve even larger negative Poisson ratios than have been realized while maintaining axial stiffness. This sort of behavior has already been explored somewhat by Saedi et al. [137] with porous SMAs. In a related point, many studies on mechanical metamaterial examine the elastic, shear, and bulk properties of a given metamaterial but few do this with any cyclic behavior. Fatigue life is an important concern in any part that is to be implemented in a real application and this area needs further development in metamaterials, particularly for AM fabricates metamaterials where defects can serve as nucleation sites for crack growth.

Smart Materials and Composites — As already discussed smart composites have been proposed, which incorporate SMA wires in an SMP matrix. The SMP has the ability to lock the composite in its deformed state and thus actuation in applications such as origami can be maintained until subsequent activation/deactivation. The usage of AM to print origami metamaterials from smart materials has not been attempted yet but could offer much more flexibility in the composite than previously achievable. In addition to smart/4D composites, tunable 4D metamaterials need to be further examined. Some authors have started to recognize the ability to tune 4D metamaterials to specific frequency ranges, rather than single frequencies, but this has only been accomplished using an SMP for band gap tuning. Applications far beyond this are possible and should be examined. For instance, an SMA-based auxetc metamaterial could be design and printed so that, depending on the thermal stimulus, it has a different volume. The thermal stimulus could be an environmental condition and the volume change could help to facilitate cooling and maintain an optimal temperature.

Multipurpose Metamaterials — A final knowledge gap in the literature identified by this work is the use of metamaterials for multiple purposes *i.e.* acoustic-thermal or mechanical-thermal metamaterials. It is very easy for one to conceive of an application where both acoustic and thermal absorption/cloaking are

both simultaneously needed (*e.g.* jet or UAV engines) or an application where a mechanical metamaterial property is desired but proper thermal conductivity or mass flow through the structure is also required (*e.g.* biomedical applications). There are no physical principle that prevents such a metamaterial from being built and with computational abilities continuing to increase, the design of such a metamaterial is feasible. It is likely that the more requirements that are placed on the design of the metamaterial, the more complex the structure would become. Thus, AM would likely be required to fabricate the designed metamaterial.

As a final concluding remark, it should be noted that in many ways, metamaterial development using AM is limited by the AM tech and not necessarily an understanding of the metamaterial structure. AM is a complex process that will take time to fully understand and exploit but with its development, metamaterials can become more complex and the bounds of possible properties can continue to be pushed.

ACKNOWLEDGMENTS

The author acknowledges the funding from the U.S. Naval Research Laboratory's Edison Memorial Graduate Training Program. The author thanks professors Alaa Elwany, Z.J. Pei, and Shiren Wang for their participation in review of this document.

REFERENCES

- D.L. Bourell, "Perspectives on Additive Manufacturing," Annual Review of Materials Research 46(1), 1–18 (2016), ISSN 1531-7331, doi:10.1146/annurev-matsci-070115-031606. URL http: //www.annualreviews.org/doi/10.1146/annurev-matsci-070115-031606.
- 2. J.P.P. Kruth, M.C. Leu, and T. Nakagawa, "Progress in additive manufacturing and rapid prototyping," *CIRP Annals - Manufacturing Technology* 47(2), 525-540 (jan 1998), ISSN 00078506, doi:10.1016/S0007-8506(07)63240-5. URL https://www.sciencedirect. com/science/article/pii/S0007850607632405?via{%}3Dihubhttps: //linkinghub.elsevier.com/retrieve/pii/S0007850607632405.
- 3. G.N. Levy, R. Schindel, and J. Kruth, "RAPID MANUFACTURING AND RAPID TOOLING WITH LAYER MANUFACTURING (LM) TECHNOLOGIES, STATE OF THE ART AND FUTURE PERSPECTIVES," CIRP Annals 52(2), 589–609 (jan 2003), ISSN 0007-8506, doi:10.1016/S0007-8506(07)60206-6. URL https://www.sciencedirect.com/science/article/pii/ S0007850607602066?via{%}3Dihub.
- C. Buchanan and L. Gardner, "Metal 3D printing in construction: A review of methods, research, applications, opportunities and challenges," *Engineering Structures* 180(October 2018), 332–348 (2019), ISSN 01410296, doi:10.1016/j.engstruct.2018.11.045. URL https://linkinghub.elsevier.com/retrieve/pii/S0141029618307958.
- S. H. Huang, P. Liu, A. Mokasdar, and L. Hou, "Additive manufacturing and its societal impact: A literature review," *International Journal of Advanced Manufacturing Technology* 67(5-8), 1191–1203 (2013), ISSN 02683768, doi:10.1007/s00170-012-4558-5.
- D. Herzog, V. Seyda, E. Wycisk, and C. Emmelmann, "Additive manufacturing of metals," Acta Mater. 117, 371–392 (2016), ISSN 13596454, doi:10.1016/j.actamat.2016.07.019. URL http: //dx.doi.org/10.1016/j.actamat.2016.07.019.

- 7. D.D. Gu, W. Meiners, K. Wissenbach, &.R. Poprawe, and R. Poprawe, "Laser additive manufacturing of metallic components: materials, processes and mechanisms Laser additive manufacturing of metallic components: materials, processes and mechanisms," International Materials Reviews 6608 (2017), ISSN 1743-2804, doi:10.1179/1743280411Y.0000000014. URL https://www.tandfonline.com/doi/ abs/10.1179/1743280411Y.0000000014{%}OAhttp://www.tandfonline.com/ action/journalInformation?journalCode=yimr20.
- T. DebRoy, H. L. Wei, J. S. Zuback, T. Mukherjee, J. W. Elmer, J. O. Milewski, A. M. Beese, A. Wilson-Heid, A. De, and W. Zhang, "Additive manufacturing of metallic components – Process, structure and properties," *Progress in Materials Science* 92, 112–224 (2018), ISSN 00796425, doi:10.1016/j.pmatsci.2017.10.001.
- 9. F. P. Melchels, M. A. Domingos, T. J. Klein, J. Malda, P. J. Bartolo, and D. W. Hutmacher, "Additive manufacturing of tissues and organs," *Progress in Polymer Science* 37(8), 1079–1104 (2012), ISSN 00796700, doi:10.1016/j.progpolymsci.2011.11.007. URL http://dx.doi.org/10.1016/j.progpolymsci.2011.11.007.
- S. V. Murphy and A. Atala, "3D bioprinting of tissues and organs," *Nature Biotechnology* 32(8), 773–785 (2014), ISSN 15461696, doi:10.1038/nbt.2958. URL http://dx.doi.org/10.1038/nbt.2958.
- D. Bourell, J. P. Kruth, M. Leu, G. Levy, D. Rosen, A. M. Beese, and A. Clare, "Materials for additive manufacturing," *CIRP Annals Manufacturing Technology* 66(2), 659–681 (2017), ISSN 17260604, doi:10.1016/j.cirp.2017.05.009. URL http://dx.doi.org/10.1016/j.cirp.2017.05.009.
- W. Du, X. Ren, C. Ma, and Z. Pei, "Ceramic binder jetting additive manufacturing : Particle coating for increasing powder sinterability and part strength," *Materials Letters* 234, 327–330 (2019), ISSN 0167-577X, doi:10.1016/j.matlet.2018.09.118. URL https://doi.org/10.1016/j.matlet.2018.09.118.
- M. K. Thompson, G. Moroni, T. Vaneker, G. Fadel, R. I. Campbell, I. Gibson, A. Bernard, J. Schulz, P. Graf, B. Ahuja, and F. Martina, "Design for Additive Manufacturing: Trends, opportunities, considerations, and constraints," *CIRP Annals - Manufacturing Technology* 65(2), 737–760 (2016), ISSN 17260604, doi:10.1016/j.cirp.2016.05.004. URL http://dx.doi.org/10.1016/j. cirp.2016.05.004.
- 14. W. Gao, Y. Zhang, D. Ramanujan, K. Ramani, Y. Chen, C. B. Williams, C. C. Wang, Y. C. Shin, S. Zhang, and P. D. Zavattieri, "The status, challenges, and future of additive manufacturing in engineering," *CAD Computer Aided Design* 69, 65–89 (2015), ISSN 00104485, doi:10.1016/j.cad.2015.04.001. URL http://dx.doi.org/10.1016/j.cad.2015.04. 001.
- 15. T. Tancogne-Dejean, A.B. Spierings, and D. Mohr, "Additively-manufactured metallic microlattice materials for high specific energy absorption under static and dynamic loading," Acta Materialia 116, 14–28 (sep 2016), ISSN 13596454, doi:10.1016/j.actamat.2016.05.054. URL http://dx.doi.org/10.1016/j.actamat.2016.05.054https://linkinghub. elsevier.com/retrieve/pii/S1359645416304153.

- 16. L. Hirt, A. Reiser, R. Spolenak, and T. Zambelli, "Additive Manufacturing of Metal Structures at the Micrometer Scale," *Advanced Materials* 29(17), 1604211 (may 2017), ISSN 09359648, doi:10.1002/adma.201604211. URL http://doi.wiley.com/10.1002/adma. 201604211.
- Q. Geng, D. Wang, P. Chen, and S. c. Chen, "Ultrafast multi-focus 3-D nano-fabrication based on two-photon polymerization," *Nature Communications* 10(1), 2179 (dec 2019), ISSN 2041-1723, doi:10.1038/s41467-019-10249-2. URL http://dx.doi.org/10.1038/s41467-019-10249-2http://www.nature.com/articles/s41467-019-10249-2.
- 18. A. Roschli, K.T. Gaul, A.M. Boulger, B.K. Post, P.C. Chesser, L.J. Love, F. Blue, and M. Borish, "Designing for Big Area Additive Manufacturing," Additive Manufacturing 25(November 2018), 275–285 (jan 2019), ISSN 22148604, doi:10.1016/j.addma.2018.11.006. URL https://doi.org/10.1016/j.addma.2018.11.006https://linkinghub. elsevier.com/retrieve/pii/S2214860418307012.
- C. Greer, A. Nycz, M. Noakes, B. Richardson, B. Post, T. Kurfess, and L. Love, "Introduction to the design rules for Metal Big Area Additive Manufacturing," *Additive Manufacturing* 27(November 2018), 159–166 (may 2019), ISSN 22148604, doi:10.1016/j.addma.2019.02.016. URL https://doi.org/10.1016/j.addma.2019.02.016https://linkinghub. elsevier.com/retrieve/pii/S2214860418309345.
- S. Tibbits, "4D Printing: Multi-Material Shape Change," Architectural Design 84(1), 116–121 (jan 2014), ISSN 00038504, doi:10.1002/ad.1710. URL http://doi.wiley.com/10.1002/ad. 1710.
- D. Raviv, W. Zhao, C. McKnelly, A. Papadopoulou, A. Kadambi, B. Shi, S. Hirsch, D. Dikovsky, M. Zyracki, C. Olguin, R. Raskar, and S. Tibbits, "Active Printed Materials for Complex Self-Evolving Deformations," *Scientific Reports* 4(1), 7422 (may 2015), ISSN 2045-2322, doi:10.1038/srep07422. URL http://www.nature.com/articles/srep07422.
- 22. F. Momeni, S. M.Mehdi Hassani.N, X. Liu, and J. Ni, "A review of 4D printing," *Materials & Design* 122, 42-79 (may 2017), ISSN 02641275, doi:10.1016/j.matdes.2017.02.068. URL http://dx.doi.org/10.1016/j.matdes.2017.02.068https://linkinghub.elsevier.com/retrieve/pii/S0264127517302034.
- 23. A. Mitchell, U. Lafont, M. Hołyńska, and C. Semprimoschnig, "Additive manufacturing A review of 4D printing and future applications," *Additive Manufacturing* 24(September 2017), 606–626 (dec 2018), ISSN 22148604, doi:10.1016/j.addma.2018.10.038. URL https://linkinghub.elsevier.com/retrieve/pii/S2214860417304013.
- 24. J. Gardan, "Smart materials in additive manufacturing: state of the art and trends," Virtual and Physical Prototyping 14(1), 1-18 (jan 2019), ISSN 1745-2759, doi:10.1080/17452759.2018.1518016. URL https://www.tandfonline.com/doi/ full/10.1080/17452759.2018.1518016.
- 25. J. Ma, B. Franco, G. Tapia, K. Karayagiz, L. Johnson, J. Liu, R. Arroyave, I. Karaman, and A. Elwany, "Spatial Control of Functional Response in 4D-Printed Active Metallic Structures," *Scientific Reports* 7(1), 46707 (may 2017), ISSN 2045-2322, doi:10.1038/srep46707. URL http://www.nature. com/articles/srep46707.

- 26. S. Miao, N. Castro, M. Nowicki, L. Xia, H. Cui, X. Zhou, W. Zhu, S.j. Lee, K. Sarkar, G. Vozzi, Y. Tabata, J. Fisher, and L.G. Zhang, "4D printing of polymeric materials for tissue and organ regeneration," *Materials Today* 20(10), 577–591 (dec 2017), ISSN 13697021, doi:10.1016/j.mattod.2017.06.005. URL https://doi.org/10.1016/j. mattod.2017.06.005https://linkinghub.elsevier.com/retrieve/pii/ S1369702117302250.
- 27. C. M. González-Henríquez, M. A. Sarabia-Vallejos, and J. Rodriguez-Hernandez, "Polymers for additive manufacturing and 4D-printing: Materials, methodologies, and biomedical applications," *Progress in Polymer Science* 94, 57–116 (jul 2019), ISSN 00796700, doi:10.1016/j.progpolymsci.2019.03.001. URL https://doi.org/10.1016/j. progpolymsci.2019.03.001https://linkinghub.elsevier.com/retrieve/ pii/S007967001830128X.
- X. Kuang, D. J. Roach, J. Wu, C. M. Hamel, Z. Ding, T. Wang, M. L. Dunn, and H. J. Qi, "Advances in 4D Printing: Materials and Applications," *Advanced Functional Materials* 29(2), 1–23 (2019), ISSN 16163028, doi:10.1002/adfm.201805290.
- 29. J.J. Schwartz and A.J. Boydston, "Multimaterial actinic spatial control 3D and 4D printing," *Nature Communications* 10(1), 791 (dec 2019), ISSN 2041-1723, doi:10.1038/s41467-019-08639-7. URL http://dx.doi.org/10.1038/s41467-019-08639-7http://www. nature.com/articles/s41467-019-08639-7.
- 30. V. G. Veselago and E. E. Narimanov, "The left hand of brightness: Past, present and future of negative index materials," *Nature Materials* **5**(10), 759–762 (2006), ISSN 14764660, doi:10.1038/nmat1746.
- N. Engheta and R. W. Ziolkowski, *Metamaterials* (John Wiley & Sons, Inc., Hoboken, NJ, USA, may 2006), ISBN 9780471784197, doi:10.1002/0471784192. URL http://doi.wiley.com/10.1002/0471784192.
- 32. Y. Liu and X. Zhang, "Metamaterials: a new frontier of science and technology," *Chemical Society Reviews* 40(5), 2494 (2011), ISSN 0306-0012, doi:10.1039/c0cs00184h. URL http://xlink.rsc.org/?DOI=c0cs00184h.
- 33. V.G. Veselago, "The electrodynamics of substances with negative epsilon amd mu," *Soviet Physics* USPEKHI **10**(4) (1968).
- 34. J. B. Pendry, "Negative Refraction Makes a Perfect Lens," *Physical Review Letters* **85**(18), 3966–3969 (2000).
- 35. D. R. Smith, W. J. Padilla, D. C. Vier, S. C. Nemat-Nasser, and S. Schultz, "Composite Medium with Simultaneously Negative Permeability and Permittivity," *Physical Review Letters* **84**(18), 4184–4187 (may 2000), ISSN 0031-9007, doi:10.1103/PhysRevLett.84.4184. URL https://link.aps.org/doi/10.1103/PhysRevLett.84.4184.
- R. A. Shelby, D. R. Smith, and S. Schultz, "Experimental Verification of a Negative Index of Refraction," *Science* 292(5514), 77–79 (apr 2001), ISSN 00368075, doi:10.1126/science.1058847. URL http://www.sciencemag.org/cgi/doi/10.1126/science.1058847.
- C. M. Soukoulis and M. Wegener, "Optical Metamaterials-More Bulky and Less Lossy," Science 330(6011), 1633-1634 (dec 2010), ISSN 0036-8075, doi:10.1126/science.1198858. URL http: //www.sciencemag.org/cgi/doi/10.1126/science.1198858.

- D. R. Smith, J. B. Pendry, and M. C. Wiltshire, "Metamaterials and negative refractive index," *Science* 305(5685), 788–792 (2004), ISSN 00368075, doi:10.1126/science.1096796.
- 39. J. B. Pendry, "Controlling Electromagnetic Fields," Science 312(5781), 1780–1782 (jun 2006), ISSN 0036-8075, doi:10.1126/science.1125907. URL http://www.sciencemag.org/cgi/doi/10.1126/science.1125907.
- 40. D. Schurig, J. J. Mock, B. J. Justice, S. A. Cummer, J. B. Pendry, A. F. Starr, and D. R. Smith, "Metamaterial Electromagnetic Cloak at Microwave Frequencies," *Science* **314**(5801), 977–980 (nov 2006), ISSN 0036-8075, doi:10.1126/science.1133628. URL http://www.sciencemag.org/ cgi/doi/10.1126/science.1133628.
- H.t. Chen, W.J. Padilla, J.M.O. Zide, A.C. Gossard, A.J. Taylor, and R.D. Averitt, "Active terahertz metamaterial devices," *Nature* 444(7119), 597–600 (nov 2006), ISSN 0028-0836, doi:10.1038/nature05343. URL http://www.nature.com/articles/nature05343.
- 42. N.I. Landy, S. Sajuyigbe, J.J. Mock, D.R. Smith, and W.J. Padilla, "Perfect Metamaterial Absorber," *Physical Review Letters* 100(20), 207402 (may 2008), ISSN 0031-9007, doi:10.1103/PhysRevLett.100.207402. URL https://link.aps.org/doi/10.1103/ PhysRevLett.100.207402.
- J. Valentine, S. Zhang, T. Zentgraf, E. Ulin-Avila, D. A. Genov, G. Bartal, and X. Zhang, "Threedimensional optical metamaterial with a negative refractive index," *Nature* 455(7211), 376–379 (sep 2008), ISSN 0028-0836, doi:10.1038/nature07247. URL http://www.nature.com/ articles/nature07247.
- 44. A. A. Zhilin and M. P. Shepilov, "Metamaterials: A new direction in materials science," *Glass Physics and Chemistry* **36**(5), 521–553 (oct 2010), ISSN 1087-6596, doi:10.1134/S1087659610050019. URL http://link.springer.com/10.1134/S1087659610050019.
- 45. C. M. Soukoulis and M. Wegener, "Past achievements and future challenges in the development of three-dimensional photonic metamaterials," *Nature Photonics* 5(9), 523–530 (sep 2011), ISSN 1749-4885, doi:10.1038/nphoton.2011.154. URL http://dx.doi.org/10.1038/nphoton. 2011.154http://www.nature.com/articles/nphoton.2011.154.
- 46. M. Wegener, "Metamaterials Beyond Optics," *Science* **342**(6161), 939–940 (nov 2013), ISSN 0036-8075, doi:10.1126/science.1246545. URL http://www.sciencemag.org/cgi/doi/10. 1126/science.1246545.
- 47. M. Kadic, T. Bückmann, R. Schittny, and M. Wegener, "Metamaterials beyond electromagnetism," *Reports on Progress in Physics* 76(12), 126501 (dec 2013), ISSN 0034-4885, doi:10.1088/0034-4885/76/12/126501. URL http://stacks.iop.org/0034-4885/76/i=12/a=126501? key=crossref.c8411a30b39d4a4d15f7e72e97540ef6.
- 48. T. Koschny, C. M. Soukoulis, and M. Wegener, "Metamaterials in microwaves, optics, mechanics, thermodynamics, and transport," *Journal of Optics* 19(8), 084005 (aug 2017), ISSN 2040-8978, doi:10.1088/2040-8986/aa7288. URL http://stacks.iop.org/2040-8986/19/i=8/a= 084005?key=crossref.a223b579cd2ae2d89e556ca7b1a2059e.
- C. Z. Fan, Y. Gao, and J. P. Huang, "Shaped graded materials with an apparent negative thermal conductivity," *Applied Physics Letters* 92(25), 251907 (jun 2008), ISSN 0003-6951, doi:10.1063/1.2951600. URL http://aip.scitation.org/doi/10.1063/1.2951600.

- 50. S. Guenneau, C. Amra, and D. Veynante, "Transformation thermodynamics: cloaking and concentrating heat flux," Optics Express 20(7), 8207 (mar 2012), ISSN 1094-4087, doi:10.1364/OE.20.008207. URL http://stacks.iop.org/0034-4885/76/i= 12/a=126501?key=crossref.c8411a30b39d4a4d15f7e72e97540ef6https: //www.osapublishing.org/oe/abstract.cfm?uri=oe-20-7-8207.
- 51. S.R. Sklan and B. Li, "Thermal metamaterials: functions and prospects," National Science Review 5(2), 138-141 (mar 2018), ISSN 2095-5138, doi:10.1093/nsr/nwy005. URL https: //academic.oup.com/nsr/article/5/2/138/4803969.
- C. Park, R. T. Haftka, and N. H. Kim, "Remarks on multi-fidelity surrogates," *Structural and Multidisciplinary Optimization* 55(3), 1029–1050 (2017), ISSN 16151488, doi:10.1007/s00158-016-1550-y.
- 53. E.M. Dede, F. Zhou, P. Schmalenberg, and T. Nomura, "Thermal Metamaterials for Heat Flow Control in Electronics," *Journal of Electronic Packaging* 140(1), 010904 (mar 2018), ISSN 1043-7398, doi:10.1115/1.4039020. URL http://electronicpackaging. asmedigitalcollection.asme.org/article.aspx?doi=10.1115/1.4039020.
- 54. J. Christensen, M. Kadic, O. Kraft, and M. Wegener, "Vibrant times for mechanical metamaterials," MRS Communications 5(3), 453-462 (sep 2015), ISSN 2159-6859, doi:10.1557/mrc.2015.51. URL https://www.cambridge.org/core/product/ identifier/S2159685915000518/type/journal{_}article.
- 55. Z. Liu, "Locally Resonant Sonic Materials," *Science* 289(5485), 1734–1736 (sep 2000), ISSN 00368075, doi:10.1126/science.289.5485.1734. URL http://www.sciencemag.org/cgi/doi/10.1126/science.289.5485.1734.
- 56. S. Chen, Y. Fan, Q. Fu, H. Wu, Y. Jin, J. Zheng, and F. Zhang, "A Review of Tunable Acoustic Metamaterials," *Applied Sciences* 8(9), 1480 (aug 2018), ISSN 2076-3417, doi:10.3390/app8091480. URL http://www.mdpi.com/2076-3417/8/9/1480.
- 57. Z.J. Wong, Y. Wang, K. O'Brien, J. Rho, X. Yin, S. Zhang, N. Fang, T.J. Yen, and X. Zhang, "Optical and acoustic metamaterials: superlens, negative refractive index and invisibility cloak," *Journal of Optics* 19(8), 084007 (aug 2017), ISSN 2040-8978, doi:10.1088/2040-8986/aa7a1f. URL http://stacks.iop.org/2040-8986/19/i=8/a=084007?key= crossref.edf2030bfb5f23e837c73092b75e6ca1.
- 58. Y. I. Bobrovnitskii and T. M. Tomilina, "Sound Absorption and Metamaterials: A Review," Acoustical Physics 64(5), 519–526 (sep 2018), ISSN 1063-7710, doi:10.1134/S1063771018040024. URL http://link.springer.com/10.1134/S1063771018040024.
- 59. S.A. Cummer, J. Christensen, and A. Alù, "Controlling sound with acoustic metamaterials," *Nature Reviews Materials* 1(3), 16001 (mar 2016), ISSN 2058-8437, doi:10.1038/natrevmats.2016.1. URL http://dx.doi.org/10.1038/natrevmats.2016.1http://www.nature.com/articles/natrevmats20161.
- M. R. Haberman and M. D. Guild, "Acoustic metamaterials," *Physics Today* 69(6), 42–48 (jun 2016), ISSN 0031-9228, doi:10.1063/PT.3.3198. URL http://physicstoday.scitation.org/ doi/10.1063/PT.3.3198.

- 61. G. Ma and P. Sheng, "Acoustic metamaterials: From local resonances to broad horizons," Science Advances 2(2), e1501595 (feb 2016), ISSN 2375-2548, doi:10.1126/sciadv.1501595. URL http://www.mdpi.com/2076-3417/8/9/1480http://advances.sciencemag. org/lookup/doi/10.1126/sciadv.1501595.
- R. Gatt, L. Mizzi, J.I. Azzopardi, K.M. Azzopardi, D. Attard, A. Casha, J. Briffa, and J.N. Grima, "Hierarchical Auxetic Mechanical Metamaterials," *Scientific Reports* 5(1), 8395 (jul 2015), ISSN 2045-2322, doi:10.1038/srep08395. URL http://www.nature.com/articles/srep08395.
- K. K. Saxena, R. Das, and E. P. Calius, "Three Decades of Auxetics Research Materials with Negative Poisson's Ratio: A Review," *Advanced Engineering Materials* 18(11), 1847–1870 (nov 2016), ISSN 14381656, doi:10.1002/adem.201600053. URL http://doi.wiley.com/10. 1002/adem.201600053.
- 64. H. M. Kolken and A. A. Zadpoor, "Auxetic mechanical metamaterials," *RSC Advances* **7**(9), 5111–5129 (2017), ISSN 20462069, doi:10.1039/c6ra27333e.
- 65. X. Ren, R. Das, P. Tran, T. D. Ngo, and Y. M. Xie, "Auxetic metamaterials and structures: a review," *Smart Materials and Structures* 27(2), 023001 (feb 2018), ISSN 0964-1726, doi:10.1088/1361-665X/aaa61c. URL http://stacks.iop.org/0964-1726/27/i=2/a=023001?key= crossref.723edd00517a5c9d7d39ee8579e85193.
- 66. M. Kadic, T. Bückmann, N. Stenger, M. Thiel, and M. Wegener, "On the practicability of pentamode mechanical metamaterials," *Applied Physics Letters* 100(19), 191901 (may 2012), ISSN 0003-6951, doi:10.1063/1.4709436. URL http://aip.scitation.org/doi/10.1063/1.4709436.
- 67. T. Bückmann, M. Thiel, M. Kadic, R. Schittny, and M. Wegener, "An elasto-mechanical unfeelability cloak made of pentamode metamaterials," *Nature Communications* 5(1), 4130 (sep 2014), ISSN 2041-1723, doi:10.1038/ncomms5130. URL http://www.nature.com/articles/ncomms5130.
- R. S. Lakes, T. Lee, A. Bersie, and Y.C. Wang, "Extreme damping in composite materials with negative-stiffness inclusions," *Nature* 410(6828), 565–567 (mar 2001), ISSN 0028-0836, doi:10.1038/35069035. URL http://www.nature.com/articles/35069035.
- 69. Z.G. Nicolaou and A.E. Motter, "Mechanical metamaterials with negative compressibility transitions," *Nature Materials* 11(7), 608-613 (jul 2012), ISSN 1476-1122, doi:10.1038/nmat3331. URL http://dx.doi.org/10.1038/nmat3331http://www.nature.com/articles/nmat3331.
- Q. Wang, J.A. Jackson, Q. Ge, J.B. Hopkins, C.M. Spadaccini, and N.X. Fang, "Lightweight Mechanical Metamaterials with Tunable Negative Thermal Expansion," *Physical Review Letters* 117(17), 175901 (oct 2016), ISSN 0031-9007, doi:10.1103/PhysRevLett.117.175901. URL https://link.aps.org/doi/10.1103/PhysRevLett.117.175901.
- 71. L. Ai and X.L. Gao, "Metamaterials with negative Poisson's ratio and non-positive thermal expansion," *Composite Structures* **162**, 70–84 (2017), ISSN 02638223, doi:10.1016/j.compstruct.2016.11.056.

- 72. A. Alderson and K. L. Alderson, "Auxetic materials," *Proceedings of the Institution of Mechanical Engineers, Part G: Journal of Aerospace Engineering* 221(4), 565–575 (apr 2007), ISSN 0954-4100, doi:10.1243/09544100JAERO185. URL http://journals.sagepub.com/doi/10.1243/09544100JAERO185.
- 73. Y. Liu and H. Hu, "A review on auxetic structures and polymeric materials," *Scientific Research and Essays* **5**(10), 1052–1063 (2015).
- 74. A. A. Zadpoor, "Mechanical meta-materials," *Materials Horizons* **3**(5), 371–381 (2016), ISSN 2051-6347, doi:10.1039/C6MH00065G. URL http://xlink.rsc.org/?D0I=C6MH00065G.
- 75. X. Yu, J. Zhou, H. Liang, Z. Jiang, and L. Wu, "Mechanical metamaterials associated with stiffness, rigidity and compressibility: A brief review," *Progress in Materials Science* 94, 114–173 (2018), ISSN 00796425, doi:10.1016/j.pmatsci.2017.12.003. URL https://doi.org/10.1016/j.pmatsci.2017.12.003.
- 76. E. Barchiesi, M. Spagnuolo, and L. Placidi, "Mechanical metamaterials: a state of the art," *Mathematics and Mechanics of Solids* 24(1), 212–234 (jan 2019), ISSN 1081-2865, doi:10.1177/1081286517735695. URL http://journals.sagepub.com/doi/10.1177/ 1081286517735695.
- 77. D. Lin, Q. Nian, B. Deng, S. Jin, Y. Hu, W. Wang, and G. J. Cheng, "Three-Dimensional Printing of Complex Structures: Man Made or toward Nature?," ACS Nano 8(10), 9710–9715 (oct 2014), ISSN 1936-0851, doi:10.1021/nn504894j. URL https://pubs.acs.org/doi/10.1021/ nn504894j.
- 78. M. Mao, J. He, X. Li, B. Zhang, Q. Lei, Y. Liu, and D. Li, "The Emerging Frontiers and Applications of High-Resolution 3D Printing," *Micromachines* 8(4), 113 (apr 2017), ISSN 2072-666X, doi:10.3390/mi8040113. URL http://www.mdpi.com/2072-666X/8/4/113.
- 79. S. Walia, C. M. Shah, P. Gutruf, H. Nili, D. R. Chowdhury, W. Withayachumnankul, M. Bhaskaran, and S. Sriram, "Flexible metasurfaces and metamaterials: A review of materials and fabrication processes at micro- and nano-scales," *Applied Physics Reviews* 2(1), 011303 (mar 2015), ISSN 1931-9401, doi:10.1063/1.4913751. URL http://dx.doi.org/10.1063/1.4913751http://aip.scitation.org/doi/10.1063/1.4913751.
- 80. G. Yoon, I. Kim, and J. Rho, "Challenges in fabrication towards realization of practical metamaterials," *Microelectronic Engineering* 163, 7–20 (sep 2016), ISSN 01679317, doi:10.1016/j.mee.2016.05.005. URL http://dx.doi.org/10.1016/j.mee.2016.05.005https://linkinghub.elsevier.com/retrieve/pii/S0167931716302337.
- 81. J. Bishop-Moser, C. Spadaccini, and C. Andres, "Metamaterials Manufacturing: Pathway to Industrial Competitiveness, 2018.
- 82. X. Zheng, H. Lee, T.H. Weisgraber, M. Shusteff, J. DeOtte, E.B. Duoss, J.D. Kuntz, M.M. Biener, Q. Ge, J.A. Jackson, S.O. Kucheyev, N.X. Fang, and C.M. Spadaccini, "Ultralight, ultrastiff mechanical metamaterials," *Science* 344(6190), 1373–1377 (jun 2014), ISSN 0036-8075, doi:10.1126/science.1252291. URL http://www.sciencemag.org/cgi/doi/10.1126/science.1252291.

- M. Deubel, G. von Freymann, M. Wegener, S. Pereira, K. Busch, and C. M. Soukoulis, "Direct laser writing of three-dimensional photonic-crystal templates for telecommunications," *Nature Materials* 3(7), 444–447 (jul 2004), ISSN 1476-1122, doi:10.1038/nmat1155. URL http://www.nature.com/articles/nmat1155.
- 84. A. Boltasseva and V.M. Shalaev, "Fabrication of optical negative-index metamaterials: Recent advances and outlook," *Metamaterials* 2(1), 1-17 (may 2008), ISSN 18731988, doi:10.1016/j.metmat.2008.03.004. URL https://linkinghub.elsevier.com/ retrieve/pii/S1873198808000066.
- 85. B. Du, J. Zhou, and L. F. Hao, "Fabrication and properties of meta-materials based on multilayer ceramic structure," *Journal of Electroceramics* 21(1-4), 165–169 (dec 2008), ISSN 1385-3449, doi:10.1007/s10832-007-9113-7. URL http://link.springer.com/10.1007/s10832-007-9113-7.
- 86. J. Wang, S. Liu, S. Guruswamy, and A. Nahata, "Injection Molding of Free-Standing, Three-Dimensional, All-Metal Terahertz Metamaterials," *Advanced Optical Materials* 2(7), 663–669 (jul 2014), ISSN 21951071, doi:10.1002/adom.201400094. URL http://doi.wiley.com/10. 1002/adom.201400094.
- L. Gao, K. Shigeta, A. Vazquez-Guardado, C. J. Progler, G. R. Bogart, J. A. Rogers, and D. Chanda, "Nanoimprinting Techniques for Large-Area Three-Dimensional Negative Index Metamaterials with Operation in the Visible and Telecom Bands," ACS Nano 8(6), 5535–5542 (jun 2014), ISSN 1936-0851, doi:10.1021/nn5015775. URL https://pubs.acs.org/doi/10.1021/nn5015775.
- 88. T. Paik, H. Yun, B. Fleury, S. h. Hong, P. S. Jo, Y. Wu, S. J. Oh, M. Cargnello, H. Yang, C. B. Murray, and C. R. Kagan, "Hierarchical Materials Design by Pattern Transfer Printing of Self-Assembled Binary Nanocrystal Superlattices," *Nano Letters* 17(3), 1387–1394 (mar 2017), ISSN 1530-6984, doi:10.1021/acs.nanolett.6b04279. URL https://pubs.acs.org/doi/10.1021/acs.nanolett.6b04279.
- S. Vignolini, N.A. Yufa, P.S. Cunha, S. Guldin, I. Rushkin, M. Stefik, K. Hur, U. Wiesner, J.J. Baumberg, and U. Steiner, "A 3D Optical Metamaterial Made by Self-Assembly," *Advanced Materials* 24(10), OP23-OP27 (mar 2012), ISSN 09359648, doi:10.1002/adma.201103610. URL http://doi.wiley.com/10.1002/adma.201103610.
- 90. J. Y. Kim, H. Kim, B. H. Kim, T. Chang, J. Lim, H. M. Jin, J. H. Mun, Y. J. Choi, K. Chung, J. Shin, S. Fan, and S. O. Kim, "Highly tunable refractive index visible-light metasurface from block copolymer self-assembly," *Nature Communications* 7(1), 12911 (dec 2016), ISSN 2041-1723, doi:10.1038/ncomms12911. URL http://dx.doi.org/10.1038/ncomms12911http://www.nature.com/articles/ncomms12911.
- 91. W. Carter, D. Erno, D. Abbott, C. Bruck, G. Wilson, J. Wolfe, D. Finkhousen, A. Tepper, and R. Stevens, "The GE aircraft engine bracket challenge: an experiment in crowdsourcing for mechanical design concepts," Proceedings of the 25th Annual International Solid Freeform Fabrication Symposium, 2014, pp. 1402–1411.
- 92. L. E. Murr, S. M. Gaytan, D. A. Ramirez, E. Martinez, J. Hernandez, K. N. Amato, P. W. Shindo, F. R. Medina, and R. B. Wicker, "Metal fabrication by additive manufacturing using laser and electron beam melting technologies," *J. Mater. Sci. Technol.* 28(1), 1–14 (2012), ISSN

10050302, doi:10.1016/S1005-0302(12)60016-4. URL http://dx.doi.org/10.1016/S1005-0302(12)60016-4.

- 93. R. Raman, B. Bhaduri, M. Mir, A. Shkumatov, M. K. Lee, G. Popescu, H. Kong, and R. Bashir, "High-Resolution Projection Microstereolithography for Patterning of Neovasculature," *Advanced Healthcare Materials* 5(5), 610–619 (mar 2016), ISSN 21922640, doi:10.1002/adhm.201500721. URL http://doi.wiley.com/10.1002/adhm.201500721.
- Z. Jiang, W. Chen, and B. J. German, "Multidisciplinary Statistical Sensitivity Analysis Considering Both Aleatory and Epistemic Uncertainties," *AIAA Journal* 54(4), 1326–1338 (2016), ISSN 0001-1452, doi:10.2514/1.j054464.
- 95. H. Cui, R. Hensleigh, H. Chen, and X. Zheng, "Additive Manufacturing and sizedependent mechanical properties of three-dimensional microarchitected, high-temperature ceramic metamaterials," *Journal of Materials Research* 33(3), 360–371 (2018), ISSN 20445326, doi:10.1557/jmr.2018.11.
- X. Zheng, W. Smith, J. Jackson, B. Moran, H. Cui, D. Chen, J. Ye, N. Fang, N. Rodriguez, T. Weisgraber, and C. M. Spadaccini, "Multiscale metallic metamaterials," *Nature Materials* 15(10), 1100–1106 (2016), ISSN 14764660, doi:10.1038/nmat4694.
- 97. J. U. Surjadi, L. Gao, H. Du, X. Li, X. Xiong, N. X. Fang, and Y. Lu, "Mechanical Metamaterials and Their Engineering Applications," *Advanced Engineering Materials* 21(3), 1800864 (mar 2019), ISSN 1438-1656, doi:10.1002/adem.201800864. URL https://onlinelibrary.wiley. com/doi/abs/10.1002/adem.201800864.
- 98. S. Hengsbach and A. D. Lantada, "Direct laser writing of auxetic structures: present capabilities and challenges," *Smart Materials and Structures* 23(8), 085033 (aug 2014), ISSN 0964-1726, doi:10.1088/0964-1726/23/8/085033. URL http://stacks.iop.org/0964-1726/23/i= 8/a=085033?key=crossref.db0e874a36aeb3367c01299918dca556.
- 99. J.A. Lewis, "Direct-write assembly of ceramics from colloidal inks," Current Opinion in Solid State and Materials Science 6(3), 245-250 (jun 2002), ISSN 13590286, doi:10.1016/S1359-0286(02)00031-1. URL https://linkinghub.elsevier.com/retrieve/pii/ S1359028602000311.
- 100. J. E. Smay, J. Cesarano, and J. A. Lewis, "Colloidal Inks for Directed Assembly of 3-D Periodic Structures," *Langmuir* 18(14), 5429–5437 (jul 2002), ISSN 0743-7463, doi:10.1021/la0257135. URL https://pubs.acs.org/doi/10.1021/la0257135.
- C. M. Spadaccini, "Mechanical Metamaterials : Design, Fabrication, and Performance," Proceedings of the Frontiers of Engineering: Reports on Leading-Edge Engineering from the 2015 Symposium (National Academies Press), 2016.
- 102. L. BESRA and M. LIU, "A review on fundamentals and applications of electrophoretic deposition (EPD)," Progress in Materials Science 52(1), 1-61 (jan 2007), ISSN 00796425, doi:10.1016/j.pmatsci.2006.07.001. URL https://linkinghub.elsevier.com/ retrieve/pii/S0079642506000387.
- A. Reiser, M. Lindén, P. Rohner, A. Marchand, H. Galinski, A.S. Sologubenko, J.M. Wheeler, R. Zenobi, D. Poulikakos, and R. Spolenak, "Multi-metal electrohydrodynamic redox 3D

printing at the submicron scale," *Nature Communications* **10**(1), 1853 (dec 2019), ISSN 2041-1723, doi:10.1038/s41467-019-09827-1. URL http://dx.doi.org/10.1038/s41467-019-09827-1http://www.nature.com/articles/s41467-019-09827-1.

- 104. M. Mohsenizadeh, F. Gasbarri, M. Munther, A. Beheshti, and K. Davami, "Additively-manufactured lightweight Metamaterials for energy absorption," *Materials and Design* 139, 521–530 (2018), ISSN 18734197, doi:10.1016/j.matdes.2017.11.037. URL https://doi.org/10.1016/j.matdes.2017.11.037.
- 105. A. Sadeqi, H. Rezaei Nejad, R.E. Owyeung, and S. Sonkusale, "Three dimensional printing of metamaterial embedded geometrical optics (MEGO)," *Microsystems & Nanoengineering* 5(1), 16 (dec 2019), ISSN 2055-7434, doi:10.1038/s41378-019-0053-6. URL http://dx.doi.org/10.1038/s41378-019-0053-6http://www.nature. com/articles/s41378-019-0053-6.
- 106. L. Kashdan, C. Conner Seepersad, M. Haberman, and P.S. Wilson, "Design, fabrication, and evaluation of negative stiffness elements using SLS," *Rapid Prototyping Journal* 18(3), 194– 200 (apr 2012), ISSN 1355-2546, doi:10.1108/13552541211218108. URL http://www. emeraldinsight.com/doi/10.1108/13552541211218108.
- 107. Y. Sha, L. Jiani, C. Haoyu, R.O. Ritchie, and X. Jun, "Design and strengthening mechanisms in hierarchical architected materials processed using additive manufacturing," *International Journal of Mechanical Sciences* 149(August), 150–163 (2018), ISSN 00207403, doi:10.1016/j.ijmecsci.2018.09.038.
- 108. S. Yuan, C. K. Chua, and K. Zhou, "3D-Printed Mechanical Metamaterials with High Energy Absorption," Advanced Materials Technologies 4(3), 1800419 (mar 2019), ISSN 2365709X, doi:10.1002/admt.201800419. URL http://doi.wiley.com/10.1002/admt. 201800419.
- 109. M.J. Mirzaali, H. Pahlavani, and A.A. Zadpoor, "Auxeticity and stiffness of random networks: Lessons for the rational design of 3D printed mechanical metamaterials," *Applied Physics Letters* 115(2), 021901 (jul 2019), ISSN 0003-6951, doi:10.1063/1.5096590. URL http://aip. scitation.org/doi/10.1063/1.5096590.
- 110. A.D. Dressler, E.W. Jost, J.C. Miers, D.G. Moore, C.C. Seepersad, and B.L. Boyce, "Heterogeneities dominate mechanical performance of additively manufactured metal lattice struts," *Additive Manufacturing* 28(June), 692–703 (aug 2019), ISSN 22148604, doi:10.1016/j.addma.2019.06.011. URL https://doi.org/10.1016/j.addma.2019.06. 011https://linkinghub.elsevier.com/retrieve/pii/S2214860418307693.
- 111. M. Wormser, F. Wein, M. Stingl, and C. Körner, "Design and Additive Manufacturing of 3D Phononic Band Gap Structures Based on Gradient Based Optimization," *Materials* 10(10), 1125 (sep 2017), ISSN 1996-1944, doi:10.3390/ma10101125. URL http://www.mdpi.com/1996-1944/10/ 10/1125.
- 112. M. Bodaghi and W. H. Liao, "4D printed tunable mechanical metamaterials with shape memory operations," *Smart Materials and Structures* 28(4), 045019 (apr 2019), ISSN 0964-1726, doi:10.1088/1361-665X/ab0b6b. URL http://stacks.iop.org/0964-1726/28/i=4/a= 045019?key=crossref.43717cff16f570a7f7eb3840dce6bd9e.

- 113. D. Isakov, Q. Lei, F. Castles, C. Stevens, C. Grovenor, and P. Grant, "3D printed anisotropic dielectric composite with meta-material features," *Materials & Design* 93, 423–430 (mar 2016), ISSN 02641275, doi:10.1016/j.matdes.2015.12.176. URL http://dx.doi.org/10.1016/j.matdes.2015.12.176https://linkinghub. elsevier.com/retrieve/pii/S0264127515310340.
- 114. Y. Tang, Y. Zhou, T. Hoff, M. Garon, and Y.F. Zhao, "Elastic modulus of 316 stainless steel lattice structure fabricated via binder jetting process," *Materials Science and Technology* **32**(7), 648–656 (may 2016), ISSN 0267-0836, doi:10.1179/1743284715Y.0000000084. URL http: //www.tandfonline.com/doi/full/10.1179/1743284715Y.0000000084.
- 115. D. Jahn, R. Eckstein, L. M. Schneider, N. Born, G. Hernandez-Sosa, J. C. Balzer, I. Al-Naib, U. Lemmer, and M. Koch, "Digital Aerosol Jet Printing for the Fabrication of Terahertz Metamaterials," *Advanced Materials Technologies* 3(2), 1–7 (2018), ISSN 2365709X, doi:10.1002/admt.201700236.
- 116. J. Akedo, "Aerosol Deposition of Ceramic Thick Films at Room Temperature: Densification Mechanism of Ceramic Layers," *Journal of the American Ceramic Society* 89(6), 1834–1839 (jun 2006), ISSN 0002-7820, doi:10.1111/j.1551-2916.2006.01030.x. URL http://doi.wiley. com/10.1111/j.1551-2916.2006.01030.x.
- 117. H. Assadi, H. Kreye, F. Gärtner, and T. Klassen, "Cold spraying A materials perspective," Acta Materialia 116, 382–407 (sep 2016), ISSN 13596454, doi:10.1016/j.actamat.2016.06.034. URL http://dx.doi.org/10.1016/j.actamat.2016.06.034https://linkinghub. elsevier.com/retrieve/pii/S1359645416304530.
- 118. C. Yang, M. Boorugu, A. Dopp, J. Ren, R. Martin, D. Han, W. Choi, and H. Lee, "4D printing reconfigurable, deployable and mechanically tunable metamaterials," *Materials Horizons* 6(6), 1244– 1250 (2019), ISSN 2051-6347, doi:10.1039/C9MH00302A. URL http://xlink.rsc.org/ ?DOI=C9MH00302A.
- 119. P. Li, D. H. Warner, A. Fatemi, and N. Phan, "Critical assessment of the fatigue performance of additively manufactured Ti 6A1 4V and perspective for future research **85**, 130–143 (2016), doi:10.1016/j.ijfatigue.2015.12.003.
- 120. M. Bodaghi, A. R. Damanpack, and W. H. Liao, "Self-expanding/shrinking structures by 4D printing," *Smart Materials and Structures* 25(10), 105034 (oct 2016), ISSN 0964-1726, doi:10.1088/0964-1726/25/10/105034. URL http://stacks.iop.org/0964-1726/25/i=10/a=105034? key=crossref.e6a45c5bfb224ff477e0748dc7e79c97.
- 121. W. Abuzaid, M. Alkhader, and M. Omari, "Experimental analysis of heterogeneous shape recovery in 4d printed honeycomb structures," *Polymer Testing* 68(February), 100-109 (jul 2018), ISSN 01429418, doi:10.1016/j.polymertesting.2018.03.050. URL https://doi.org/10.1016/j.polymertesting.2018.03.050https: //linkinghub.elsevier.com/retrieve/pii/S0142941818303179.
- 122. M. Bodaghi, A. Damanpack, and W. Liao, "Adaptive metamaterials by functionally graded 4D printing," *Materials & Design* 135, 26–36 (dec 2017), ISSN 02641275, doi:10.1016/j.matdes.2017.08.069. URL https://doi.org/10.1016/j. matdes.2017.08.069https://linkinghub.elsevier.com/retrieve/pii/ S0264127517308237.

- 123. X. Wu, Y. Su, and J. Shi, "Perspective of additive manufacturing for metamaterials development," Smart Materials and Structures 28(9), 093001 (sep 2019), ISSN 0964-1726, doi:10.1088/1361-665X/ab2eb6. URL https://iopscience.iop.org/article/10.1088/1361-665X/ ab2eb6.
- 124. C. R. Garcia, *3D PRINTED SPATIALLY VARIANT ANISOTROPIC METAMATERIALS*, PhD thesis (The University of Texas at El Paso, 2014).
- 125. L. Wu, B. Li, and J. Zhou, "Isotropic Negative Thermal Expansion Metamaterials," ACS Applied Materials & Interfaces 8(27), 17721-17727 (jul 2016), ISSN 1944-8244, doi:10.1021/acsami.6b05717. URL https://pubs.acs.org/doi/10.1021/acsami. 6b05717.
- 126. L. Sanchis, V. M. García-Chocano, R. Llopis-Pontiveros, A. Climente, J. Martínez-Pastor, F. Cervera, and J. Sánchez-Dehesa, "Three-Dimensional Axisymmetric Cloak Based on the Cancellation of Acoustic Scattering from a Sphere," *Physical Review Letters* 110(12), 124301 (mar 2013), ISSN 0031-9007, doi:10.1103/PhysRevLett.110.124301. URL https://link.aps.org/doi/10.1103/PhysRevLett.110.124301.
- 127. I. M. H. Raza, *Additive Manufacturing of Locally Resonant Metamaterials*, PhD thesis (Imperial College London, 2017).
- 128. K. H. Matlack, A. Bauhofer, S. Krödel, A. Palermo, and C. Daraio, "Composite 3Dprinted metastructures for low-frequency and broadband vibration absorption," *Proceedings* of the National Academy of Sciences 113(30), 8386–8390 (jul 2016), ISSN 0027-8424, doi:10.1073/pnas.1600171113. URL http://www.pnas.org/lookup/doi/10.1073/ pnas.1600171113.
- 129. A. Amendola, C. Smith, R. Goodall, F. Auricchio, L. Feo, G. Benzoni, and F. Fraternali, "Experimental response of additively manufactured metallic pentamode materials confined between stiffening plates," *Composite Structures* 142, 254–262 (may 2016), ISSN 02638223, doi:10.1016/j.compstruct.2016.01.091. URL http://dx.doi.org/10.1016/j. compstruct.2016.01.091https://linkinghub.elsevier.com/retrieve/pii/ S0263822316001148.
- 130. R. Hedayati, A. M. Leeflang, and A. A. Zadpoor, "Additively manufactured metallic pentamode meta-materials," *Applied Physics Letters* 110(9), 091905 (feb 2017), ISSN 0003-6951, doi:10.1063/1.4977561. URL http://dx.doi.org/10.1063/1.4977561http://aip. scitation.org/doi/10.1063/1.4977561.
- 131. R. Schittny, T. Bückmann, M. Kadic, and M. Wegener, "Elastic measurements on macroscopic threedimensional pentamode metamaterials," *Applied Physics Letters* 103(23), 231905 (dec 2013), ISSN 0003-6951, doi:10.1063/1.4838663. URL http://aip.scitation.org/doi/10.1063/1. 4838663.
- 132. D. Chen and X. Zheng, "Multi-material Additive Manufacturing of Metamaterials with Giant, Tailorable Negative Poisson's Ratios," *Scientific Reports* 8(1), 9139 (dec 2018), ISSN 2045-2322, doi:10.1038/s41598-018-26980-7. URL http://www.nature.com/articles/s41598-018-26980-7.

- 133. L. R. Meza, A. J. Zelhofer, N. Clarke, A. J. Mateos, D. M. Kochmann, and J. R. Greer, "Resilient 3D hierarchical architected metamaterials," *Proceedings of the National Academy of Sciences* 112(37), 11502–11507 (sep 2015), ISSN 0027-8424, doi:10.1073/pnas.1509120112. URL http://www.pnas.org/lookup/doi/10.1073/pnas.1509120112.
- 134. J. Bauer, L. R. Meza, T. A. Schaedler, R. Schwaiger, X. Zheng, and L. Valdevit, "Nanolattices: An Emerging Class of Mechanical Metamaterials," *Advanced Materials* 29(40), 1701850 (oct 2017), ISSN 09359648, doi:10.1002/adma.201701850. URL http://doi.wiley.com/10.1002/adma.201701850.
- 135. Q. Zhang, K. Zhang, and G. Hu, "Smart three-dimensional lightweight structure triggered from a thin composite sheet via 3D printing technique," *Scientific Reports* 6(1), 22431 (apr 2016), ISSN 2045-2322, doi:10.1038/srep22431. URL http://dx.doi.org/10.1038/srep22431http://www.nature.com/articles/srep22431.
- 136. M. Taheri Andani, S. Saedi, A. S. Turabi, M. Karamooz, C. Haberland, H. E. Karaca, and M. Elahinia, "Mechanical and shape memory properties of porous Ni 50.1 Ti 49.9 alloys manufactured by selective laser melting," *Journal of the Mechanical Behavior of Biomedical Materials* 68(January), 224–231 (apr 2017), ISSN 17516161, doi:10.1016/j.jmbbm.2017.01.047. URL http://dx.doi.org/10.1016/j.jmbbm.2017.01.047https://linkinghub. elsevier.com/retrieve/pii/S175161611730053X.
- 137. S. Saedi, S. E. Saghaian, A. Jahadakbar, N. S. Moghaddam, M. T. Andani, S. M. Saghaian, Y. C. Lu, M. Elahinia, and H. E. Karaca, "Shape memory response of porous NiTi shape memory alloys fabricated by selective laser melting," *Journal of Materials Science: Materials in Medicine* (2018), ISSN 1573-4838, doi:10.1007/s10856-018-6044-6. URL http://dx.doi.org/10.1007/s10856-018-6044-6.
- 138. L. Yin, J. Doyhamboure-Fouquet, X. Tian, and D. Li, "Design and characterization of radar absorbing structure based on gradient-refractive-index metamaterials," *Composites Part B: Engineering* 132, 178–187 (jan 2018), ISSN 13598368, doi:10.1016/j.compositesb.2017.09.003. URL https://doi.org/10.1016/j.compositesb.2017.09.003https: //linkinghub.elsevier.com/retrieve/pii/S1359836816324416.
- J. Liu, B. Jalalahmadi, Y. B. Guo, M. P. Sealy, and N. Bolander, "A review of computational modeling in powder-based additive manufacturing for metallic part qualification," *Rapid Prototyping Journal* 24(8), 1245–1264 (2018), ISSN 13552546, doi:10.1108/RPJ-04-2017-0058.
- 140. D. Han, C. Yang, N.X. Fang, and H. Lee, "Rapid multi-material 3D printing with projection micro-stereolithography using dynamic fluidic control," *Additive Manufacturing* 27(April), 606–615 (may 2019), ISSN 22148604, doi:10.1016/j.addma.2019.03.031. URL https://doi.org/10.1016/j.addma.2019.03.031https://linkinghub. elsevier.com/retrieve/pii/S2214860419301666.
- 141. E. A. Peraza-Hernandez, D. J. Hartl, R. J. Malak Jr, and D. C. Lagoudas, "Origami-inspired active structures: a synthesis and review," *Smart Materials and Structures* 23(9), 094001 (sep 2014), ISSN 0964-1726, doi:10.1088/0964-1726/23/9/094001. URL http://stacks.iop.org/0964-1726/23/i=9/a=094001?key=crossref.6226c53c3ab2492e5686c0d4be08ea2e.

- 142. R. Saunders, D. Hartl, R. Malak, and D. Lagoudas, "Design and Analysis of a Self-Folding SMA-SMP Composite Laminate," Proceedings of the Volume 5B: 38th Mechanisms and Robotics Conference (American Society of Mechanical Engineers), aug 2014. ISBN 978-0-7918-4637-7, doi:10.1115/DETC2014-35151. URL https://doi.org/10.1115/DETC2014-35151https://asmedigitalcollection.asme.org/IDETC-CIE/proceedings/ IDETC-CIE2014/46377/Buffalo, NewYork, USA/257707.
- 143. B. Wang, T. Chen, and A. Xu, "Gaussian process regression with functional covariates and multivariate response," *Chemometrics and Intelligent Laboratory Systems* 163(January), 1–6 (2017), ISSN 18733239, doi:10.1016/j.chemolab.2017.02.001. URL http://dx.doi.org/10.1016/ j.chemolab.2017.02.001.
- 144. O. K. Johnson and C. Kurniawan, "An efficient algorithm for generating diverse microstructure sets and delineating properties closures," *Acta Materialia* **147**, 313–321 (2018), ISSN 13596454, doi:10.1016/j.actamat.2018.01.004.
- 145. K. Wang, Y. Zhao, Y. h. Chang, Z. Qian, C. Zhang, B. Wang, M. A. Vannan, and M. j. Wang, "Controlling the mechanical behavior of dual-material 3D printed meta-materials for patientspecific tissue-mimicking phantoms," *Materials & Design* 90, 704–712 (jan 2016), ISSN 02641275, doi:10.1016/j.matdes.2015.11.022. URL http://dx.doi.org/10.1016/ j.matdes.2015.11.022https://linkinghub.elsevier.com/retrieve/pii/ S0264127515307644.
- 146. K. Wang, C. Wu, Z. Qian, C. Zhang, B. Wang, and M. A. Vannan, "Dual-material 3D printed metamaterials with tunable mechanical properties for patient-specific tissue-mimicking phantoms," *Additive Manufacturing* 12, 31–37 (oct 2016), ISSN 22148604, doi:10.1016/j.addma.2016.06.006. URL http://dx.doi.org/10.1016/j.addma.2016.06.006https:// linkinghub.elsevier.com/retrieve/pii/S221486041630118X.
- 147. J. Chen, K. Wang, C. Zhang, and B. Wang, "An efficient statistical approach to design 3D-printed metamaterials for mimicking mechanical properties of soft biological tissues," Additive Manufacturing 24(August), 341–352 (dec 2018), ISSN 22148604, doi:10.1016/j.addma.2018.10.007. URL https://doi.org/10.1016/j.addma.2018.10. 007https://linkinghub.elsevier.com/retrieve/pii/S2214860418305773.
- 148. H.M.A. Kolken, S. Janbaz, S.M.A. Leeflang, K. Lietaert, H.H. Weinans, and A.A. Zadpoor, "Rationally designed meta-implants: a combination of auxetic and conventional meta-biomaterials," *Materials Horizons* 5(1), 28–35 (2018), ISSN 2051-6347, doi:10.1039/C7MH00699C. URL http://xlink.rsc.org/?DOI=C7MH00699C.